# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

# FORM 8-K

# **CURRENT REPORT**

# Pursuant To Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported) November 1, 2006

# TC PipeLines, LP

(Exact name of registrant as specified in its charter)

**Delaware** (State or other jurisdiction

of incorporation)

000-26091 (Commission File Number) **52-2135448** (IRS Employer Identification No.)

**110 Turnpike Road, Suite 203 Westborough, Massachusetts** (Address of principal executive offices)

**01581** (Zip Code)

Registrant's telephone number, including area code (508) 871-7046

Not Applicable

(Former name or former address if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

o Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

## Item 1.01 <u>Entry into a Material Definitive Agreement</u>.

On November 1, 2006, TC Tuscarora Intermediate Limited Partnership, a wholly-owned subsidiary of TC PipeLines, LP (the "Partnership"), entered into a definitive agreement (the "Purchase Agreement") with Tuscarora Gas Pipeline Company ("TGPC"), a wholly-owned subsidiary of Sierra Pacific Resources, to acquire an additional 49% general partnership interest in Tuscarora Gas Transmission Company ("TGTC"), a Nevada general partnership, for \$98 million, subject to certain closing adjustments. The Partnership will also indirectly assume approximately \$37 million of debt of TGTC. The transaction is expected to close in the fourth quarter 2006, subject to regulatory approvals and other closing conditions. On closing, the Partnership's interest in TGTC will increase to 98% from 49%. TransCanada Corporation ("TransCanada"), the parent company of TC PipeLines GP, Inc. and the sole general partner of the Partnership, indirectly holds a 1% ownership interest in TGTC. Pursuant to a second agreement (the "Option Agreement"), for approximately one year after closing, the Partnership will have an option to purchase the remaining 1% interest in TGTC from TGPC for \$2 million, subject to certain closing adjustments.

TransCanada will become the operator of TGTC, which is currently operated by TGPC. TransCanada has provided gas control services for the TGTC pipeline system since late 2002.

This acquisition is expected to be accretive to the Partnership's cash flow based on anticipated financing with a combination of bank debt and equity.

The foregoing descriptions of the Purchase Agreement and the Option Agreement do not purport to be complete and are qualified by the Purchase Agreement and the Option Agreement, respectively. Copies of those agreements are attached as exhibits to this Form 8-K.

The Partnership's press release regarding the acquisition of the additional general partnership interest in TGTC is attached as an exhibit to this Form K.

## Item 9.01 <u>Financial Statements and Exhibits</u>.

8-K.

(d) <u>Exhibits</u>.

- 2.1 General Partnership Interest Purchase Agreement dated as of November 1, 2006 by and between Tuscarora Gas Pipeline Co. and TC Tuscarora Intermediate Limited Partnership.\*
- 2.2 General Partner Interest Holder Agreement dated as of November 1, 2006 by and between Tuscarora Gas Pipeline Co. and TC Tuscarora Intermediate Limited Partnership.

2

## SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

**TC PipeLines, LP** By: TC PipeLines GP, Inc., its general partner

By: /s/ Amy W. Leong Amy W. Leong Controller

Dated: November 7, 2006

3

Exhibit No.	Description
2.1	General Partnership Interest Purchase Agreement dated as of November 1, 2006 by and between Tuscarora Gas Pipeline Co. and TC Tuscarora Intermediate Limited Partnership.*
2.2	General Partner Interest Holder Agreement dated as of November 1, 2006 by and between Tuscarora Gas Pipeline Co. and TC Tuscarora Intermediate Limited Partnership.
99.1	Press Release dated November 2, 2006.

<sup>\*</sup> The schedules and exhibits to this agreement, as set forth in the Table of Contents of the agreement, have not been filed herewith pursuant to Item 601(b)(2) of Regulation S-K. The Partnership agrees to furnish any omitted materials to the Securities and Exchange Commission upon request.

4

<sup>99.1</sup> Press Release, dated November 2, 2006.

<sup>\*</sup> The schedules and exhibits to this agreement, as set forth in the Table of Contents of the agreement, have not been filed herewith pursuant to Item 601(b)(2) of Regulation S-K. The Partnership agrees to furnish any omitted materials to the Securities and Exchange Commission upon request.

## GENERAL PARTNERSHIP INTEREST

## PURCHASE AGREEMENT

# BY AND BETWEEN

# TUSCARORA GAS PIPELINE CO.

AND

# TC TUSCARORA INTERMEDIATE LIMITED PARTNERSHIP

November 1, 2006

## **INDEX OF DEFINED TERMS**

	Page
ΑΔΑ	39
AAA Affiliate	1
Agreement	1
Balance Sheet	2
Balance Sheet Date	2
Base Claim	35
Basket	35
Business	2
Business Day	2
Buyer	1
Buyer Claim	35
Buyer Disclosure Schedule	2
Buyer Indemnified Parties	34
Buyer Required Regulatory Approvals	24
Ceiling	35
CERCLA	2
Claim Notice	36
Closing	15
Closing Consideration	11
Closing Date	15
Code	3
Confidentiality Agreement	3
Contract	3
Current Assets	3
Current Liabilities	3
Dispute	39
Dispute Notice	13
Encumbrances	3
Environmental Laws	4
Environmental Permits	19
Estimated CapEx Amount	11
Estimated Cash Amount	11
Estimated Closing Adjustment	12
Estimated Debt Repayment Amount	11
Estimated Net Working Capital	11
FERC	4
Final CapEx Amount	14
Final Cash Amount	14
Final Closing Adjustment	14
Final Closing Balance Sheet	14
Final Debt Repayment Amount	14
Final Net Working Capital	14

**Financial Statements** GAAP **Governmental Authority Hazardous Substances** HSR Act **Indemnified Party Indemnifying Party** 

Independent Accounting Firm	5
Initial Purchase Price	11
Intellectual Property	6
Interest	1
Interest Holder Agreement	1
Interim Balance Sheet	6
Losses	34
Material Adverse Effect	6
Material Adverse Effect	
Material Contract Material Contracts	21 7
Material Leases Net Working Capital	777
Net Working Capital New York Courts	, 40
Notice Period	36
Operating Agreement	7
Partnership De teorebie Armenet	1
Partnership Agreement	7
Permits	22
Permitted Encumbrances	8
Person	8
Preliminary CapEx Amount	12
Preliminary Cash Amount	12
Preliminary Closing Balance Sheet	12
Preliminary Debt Repayment Amount	12
Preliminary Net Working Capital	12
Preliminary Statement	12
Prime Rate	9
Purchase Price	11
Real Property	18
Reference Net Working Capital	9
Representatives	9
Rules	39
Seller	1
Seller Claim	35
Seller Disclosure Schedule	9
Seller Indemnified Parties	35
Seller Required Regulatory Approvals	17
Seller's Knowledge	9
Software	10

2

Subsidiary 10 **Survival Period** 34 Tax 10 **Tax Return** 10 **Taxing Authority** 10 TCPL 10 **Termination Date** 33 **Third Party Claim** 36 **Transfer Taxes** 10

## GENERAL PARTNERSHIP INTEREST PURCHASE AGREEMENT

This GENERAL PARTNERSHIP INTEREST PURCHASE AGREEMENT, dated as of November 1, 2006 (this "**Agreement**"), is made and entered into by and between Tuscarora Gas Pipeline Co., a Nevada corporation ("**Seller**"), and TC Tuscarora Intermediate Limited Partnership, a Delaware limited partnership ("**Buyer**").

WHEREAS, Seller owns a 50% general partnership interest in Tuscarora Gas Transmission Company, a Nevada general partnership (the "**Partnership**");

WHEREAS, Seller desires to sell a 49% general partnership interest in the Partnership (the "Interest");

WHEREAS, concurrently with the execution of this Agreement, Seller and Buyer are entering into a General Partnership Interest Holder Agreement relating to Buyer's option to purchase an additional 1% general partnership interest in the Partnership from Seller, on the terms and subject to the conditions contained therein (the "Interest Holder Agreement"); and

WHEREAS, on the terms and subject to the conditions contained in this Agreement, Buyer desires to purchase the Interest from Seller, and Seller desires to sell the Interest to Buyer;

NOW THEREFORE, in consideration of the mutual covenants, representations, warranties and agreements contained herein, and of other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the parties hereto hereby agree as follows:

#### ARTICLE I DEFINITIONS

1.1 <u>Definitions</u>. As used in this Agreement, the following terms have the meanings specified or referred to in this Section 1.1:

(1) "AAA" shall have the meaning set forth in Section 10.5(a).

(2) "Affiliate" means a Person that directly, or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with, a specified Person. A Person shall be deemed to control another Person if such first Person possesses, directly or indirectly, the power to direct, or cause the direction of, the management and policies of such other Person, whether through the ownership of voting securities, by Contract or otherwise.

- (3) "Agreement" shall have the meaning set forth in the preamble to this Agreement.
- (4) **"Balance Sheet**" means the audited balance sheet of the Partnership as of December 31, 2005 included in the Financial
- Statements.
- (5) **"Balance Sheet Date**" means December 31, 2005.
- (6) **"Base Claim**" shall have the meaning set forth in Section 9.2(b).
- (7) **"Basket**" shall have the meaning set forth in Section 9.2(b).
- (8) "Business" means the business of the Partnership as conducted on the date of this Agreement.

(9) **"Business Day**" means any day other than Saturday, Sunday and any day which is a legal holiday or a day on which banking institutions in the State of New York are authorized by law or other governmental action to close.

- (10) **"Buyer**" shall have the meaning set forth in the preamble to this Agreement.
- (11) **"Buyer Claim**" shall have the meaning set forth in Section 9.2(b).
- (12) "Buyer Disclosure Schedule" means the disclosure schedule of Buyer referred to in, and delivered pursuant to, this Agreement.
- (13) **"Buyer Indemnified Parties**" shall have the meaning set forth in Section 9.2(a).
- (14) **"Buyer Required Regulatory Approvals**" shall have the meaning set forth in Section 5.3(b).
- (15) "Ceiling" shall have the meaning set forth in Section 9.2(b).

(16) "CERCLA" means the Comprehensive Environmental Response, Compensation and Liability Act of 1980, 42 U.S.C. §9601, et

seq., as amended.

- (17) "Claim Notice" shall have the meaning set forth in Section 9.4(a).
- (18) **"Closing**" shall have the meaning set forth in Section 3.1.
- (19) **"Closing Consideration**" shall have the meaning set forth in Section 2.2.

- (20) **"Closing Date**" shall have the meaning set forth in Section 3.1.
- (21) "Code" means the Internal Revenue Code of 1986, as amended.
- (22) "Confidentiality Agreement" means the Confidentiality Agreement, dated July 20, 2006 between Sierra Pacific Resources and an

Affiliate of Buyer.

(23) "Contract" means any written contract, agreement, indenture, note, bond, mortgage, loan, instrument, lease or license.

(24) "**Current Assets**" with respect to the Partnership, means, as of the applicable date, without duplication, the sum of the following items, each as set forth on the relevant balance sheet of the Partnership, (i) accounts receivable, (ii) other current assets and (iii) prepaid expenses (excluding, for the avoidance of doubt, cash and cash equivalents (including money market funds) and customer deposits), in each case, in accordance with GAAP and as such terms are used in, and calculated on a basis consistent with the Interim Balance Sheet.

(25) "**Current Liabilities**" with respect to the Partnership, means, as of the applicable date, without duplication, the sum of the following items, each as set forth on the relevant balance sheet of the Partnership, (i) accounts payable, (ii) payable to partners, (iii) accrued taxes other than income taxes, (iv) accrued interest and (v) current portion of long-term debt (excluding, for the avoidance of doubt, customer deposits (which are referred to as "other accrued liabilities" in the Interim Balance Sheet)), in each case, in accordance with GAAP and as such terms are used in, and calculated on a basis consistent with the Interim Balance Sheet.

(26) **"Dispute**" shall have the meaning set forth in Section 10.5(a).

(27) **"Dispute Notice**" shall have the meaning set forth in Section 2.3(e).

(28) **"Encumbrances**" means any mortgages, deeds of trust, pledges, liens, security interests, conservation easements, deed restrictions, charges and other encumbrances, other than any Permitted Encumbrances.

(29) **"Environmental Laws**" means all federal, state and local laws, including the common law, regulations, rules, ordinances, codes, decrees, judgments, directives, or judicial or administrative orders relating to pollution or protection of the environment, natural resources or human health and safety, including laws relating to Hazardous Substances (including ambient air, surface water, groundwater, land, surface and subsurface strata) or otherwise relating to the manufacture, processing, distribution, use, treatment, storage, transport or

3

handling of Hazardous Substances, laws relating to record keeping, notification, disclosure and reporting requirements respecting Hazardous Substances, and laws relating to the management and use of natural resources.

(30) **"Environmental Permits**" shall have the meaning set forth in Section 4.13(a).

(31) "Estimated CapEx Amount" shall have the meaning set forth in Section 2.3(a)(iii).

(32) "Estimated Cash Amount" shall have the meaning set forth in Section 2.3(a)(i).

(33) **"Estimated Closing Adjustment**" shall have the meaning set forth in Section 2.3(b).

(34) **"Estimated Debt Repayment Amount**" shall have the meaning set forth in Section 2.3(a)(iv).

(35) "Estimated Net Working Capital" shall have the meaning set forth in Section 2.3(a)(ii).

(36) "FERC" means the Federal Energy Regulatory Commission or any successor thereto.

(37) **"Final CapEx Amount**" shall have the meaning set forth in Section 2.3(g).

(38) "Final Cash Amount" shall have the meaning set forth in Section 2.3(g).

(39) **"Final Closing Adjustment**" shall have the meaning set forth in Section 2.3(h).

(40) "Final Closing Balance Sheet" shall have the meaning set forth in Section 2.3(g).

(41) "Final Debt Repayment Amount" shall have the meaning set forth in Section 2.3(g).

(42) "Final Net Working Capital" shall have the meaning set forth in Section 2.3(g).

(43) **"Financial Statements**" means (i) the Balance Sheet and the audited statement of income, partners' capital and cash flows of the Partnership for the fiscal year ended December 31, 2005 and (ii) the Interim Balance Sheet and the unaudited statement of income, partners' capital and cash

4

flows of the Partnership for the six (6) months ended June 30, 2006, including the notes thereto.

(44) "GAAP" means generally accepted accounting principles as used in the United States in effect from time to time.

(45) **"Governmental Authority**" means any executive, legislative, judicial, regulatory or administrative agency, body, commission, department, board, court, tribunal, arbitrating body or authority of the United States or any foreign country, or any state, local or other governmental subdivision thereof.

(46) **"Hazardous Substances**" means (i) any petrochemical or petroleum products, radioactive materials, asbestos in any form that is or could become friable, urea formaldehyde foam insulation and transformers or other equipment that contain dielectric fluid which may contain levels of

polychlorinated biphenyls, (ii) any chemicals, materials or substances defined as or included in the definition of "hazardous substances," "hazardous wastes," "hazardous materials," "restricted hazardous materials," "extremely hazardous substances," "toxic substances," "contaminants" or "pollutants" or words of similar meaning and regulatory effect under Environmental Laws or (iii) any other chemical, material or substance, exposure to which is prohibited, limited or regulated by or which could give rise to liability under any applicable Environmental Law.

- (47) "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.
- (48) **"Indemnified Party**" shall have the meaning set forth in Section 9.4(a).
- (49) "Indemnifying Party" shall have the meaning set forth in Section 9.4(a).

(50) **"Independent Accounting Firm**" means a mutually acceptable nationally recognized firm of independent chartered accountants that has not provided services to either Seller or Buyer or any of their Affiliates in the preceding three (3) years, or if no such firm is available and willing to serve, then a mutually acceptable expert in public accounting, in each case, upon which Buyer and Seller shall have mutually agreed.

(51) "Initial Purchase Price" shall have the meaning set forth in Section 2.2.

(52) **"Intellectual Property**" means all: (i) patents and patent applications, registrations and disclosures and all related continuations, divisionals, continuations-in-part, reissues, reexaminations, utility models, certificates of invention and design patents, (ii) trademarks, service marks, trade

5

dress, logos, corporate names, trade names and internet domain names, together with the goodwill associated with any of the foregoing, and all applications and registrations therefor, (iii) copyrights and registrations and applications therefor, copyrightable works of authorship and moral rights, (iv) confidential and proprietary information, including trade secrets, discoveries, concepts, ideas, research and development, financial, marketing and business data, pricing and cost information, business and marketing plans, algorithms, know-how, formulae, inventions (whether or not patentable), processes, techniques, technical data, designs, drawings, specifications, databases, and customer and supplier lists and information, in each case to the extent confidential and proprietary, excluding any rights in respect of any of the items described in this clause (iv) that comprise or are protected by patents or otherwise by clauses (i)-(iii) and (v) of this definition, and (v) Software.

(53) **"Interest**" shall have the meaning set forth in the recitals to this Agreement.

(54) "Interest Holder Agreement" shall have the meaning set forth in the recitals to this Agreement.

(55) **"Interim Balance Sheet**" means that certain unaudited balance sheet of the Partnership as at June 30, 2006 included in the Financial Statements.

(56) "**Material Adverse Effect**" means a material adverse effect on the business, results of operations, ownership, operation, or financial condition of the Partnership or the Business taken as a whole, or that materially impedes the ability of Seller to consummate the transactions contemplated by this Agreement, except for any such effect or impediment (to the extent, in the case of clauses (i), (ii) or (iii), such effect or impediment does not have a disproportionate impact on the Partnership or the Business relative to other entities operating similar businesses) arising out of or relating to (i) any change or effect resulting from the general state of the industries in which the Partnership operates (including (A) changes in pricing levels, (B) changes in the international, national, regional or local wholesale or retail markets for natural gas, (C) changes in the North American, national, regional or local interstate natural gas pipeline systems, or (D) changes in applicable laws, rules, regulations or decisions of the FERC or judgments, orders or decrees of courts affecting the interstate natural gas transmission industry, or rate orders, motions, complaints or other actions affecting the Partnership), (ii) any change or effect resulting from changes in the international, national, regional or local markets for any supplies used by the Business, (iii) any change or effect resulting from changes in general economic, political or business conditions (including changes in interest rates or debt, equity, financial, banking or currency markets), (iv) any change or effect resulting from any change in GAAP, (v) any change or effect resulting from the negotiation, execution, announcement, pendency or consummation of the transactions contemplated by this Agreement, including the impact thereof on relationships,

6

contractual or otherwise, with customers, suppliers, distributors, partners, joint owners or venturers, or employees, other than Sierra Pacific Resources and its Affiliates, (vi) any change or effect resulting from any action taken by Seller, the Partnership, Buyer or any of their respective Representatives or Affiliates or other action required, contemplated or permitted by this Agreement or consented to by Buyer, (vii) any change or effect resulting from acts of war, armed hostilities or terrorism, (viii) any change or effect resulting from changes in weather or climate, (ix) any materially adverse change in or effect on the Business which is cured (including by the payment of money) before the Termination Date, or (x) any circumstance, matter or condition described in the Seller Disclosure Schedule.

(57) "Material Contracts" shall have the meaning set forth in Section 4.15 of this Agreement

(58) **"Material Leases**" shall mean, as of the date of this Agreement, all leases under which the Partnership is a lessee, lessor or under which the Partnership otherwise has any interest (i) with annual payments greater than \$250,000 or (ii) which are otherwise material to the Business.

- (59) "Net Working Capital" means (i) Current Assets minus (ii) Current Liabilities.
- (60) "New York Courts" shall have the meaning set forth in Section 10.5(e).

(61) **"Operating Agreement**" means the Operating Agreement, dated as of October 12, 1995, by and between the Partnership and Tuscarora Gas Operating Company, as amended.

(62) "Partnership" shall have the meaning set forth in the recitals to this Agreement.

(63) **"Partnership Agreement**" means the Tuscarora Gas Transmission Company General Partnership Agreement, dated as of June 11, 1993, by and between Seller and TCPL, as amended by that First Amendment to General Partnership Agreement, dated as of September 1, 2000, as amended by that Second Amendment to General Partnership Agreement, dated as of December 17, 2003, and as may be further amended from time to time.

(64) "**Permits**" shall have the meaning set forth in Section 4.17.

(65) **"Permitted Encumbrances**" means (i) all exceptions, restrictions, easements, covenants, charges, permits, servitudes, rights of way and Encumbrances of record or that are set forth in an applicable FERC project license or title insurance policy, provided that Seller has provided or made available a copy of such license or insurance policy to Buyer, (ii) all matters of record and any state of facts that a current survey or inspection of the Real Property would disclose and which do not materially and adversely affect the

7

ability of the Partnership to conduct the Business as presently conducted, (iii) mortgages, liens, pledges, charges, Encumbrances and restrictions incurred in connection with the Partnership's purchase of properties and assets after the Balance Sheet Date securing all or a portion of the purchase price therefor, (iv) statutory Encumbrances for Taxes, assessments or other governmental charges not yet due and payable or that may be subsequently paid without penalty or interest or that are being contested in good faith in (if then appropriate) appropriate proceedings, (v) mechanics', carriers', workers', repairman's, materialman's, warehousemen's, employees', landlord's, construction and other similar Encumbrances arising or incurred in the ordinary course of business consistent with past practice relating to obligations that are not yet due and payable or that are being contested in good faith in (if then appropriate) appropriate proceedings, (v) requirements and restrictions under zoning, planning, building, entitlement, conservation, land use and environmental laws and regulations (including municipal bylaws and permits, consents and authorizations under such laws and regulations), and development, site plan, subdivision or other agreements with municipalities which do not materially and adversely affect the ability of the Partnership to conduct the Business as presently conducted, (vii) Encumbrances contemplated by or set forth in the Partnership Agreement or the Operating Agreement, (viii) the rights of lessors, lessees, licensors and licensees under leases or licenses of the Real Property, and (ix) such other liens, defects, irregularities, imperfections in or failure of title, charges, easements, restrictions and other Encumbrances which would not, individually or in the aggregate, have a Material Adverse Effect.

(66) **"Person**" means any individual, partnership, joint venture, corporation, limited liability company, limited liability partnership, trust, unincorporated organization or Governmental Authority or any department or agency thereof.

- (67) **"Preliminary CapEx Amount**" shall have the meaning set forth in Section 2.3(c)(iv).
- (68) "Preliminary Cash Amount" shall have the meaning set forth in Section 2.3(c)(ii).
- (69) "Preliminary Closing Balance Sheet" shall have the meaning set forth in Section 2.3(c)(i).
- (70) "Preliminary Debt Repayment Amount" shall have the meaning set forth in Section 2.3(c)(v).
- (71) **"Preliminary Net Working Capital**" shall have the meaning set forth in Section 2.3(c)(iii).
- (72) "Preliminary Statement" shall have the meaning set forth in Section 2.3(c).

8

(73) **"Prime Rate**" means the U.S. prime rate of interest published in the "Money Rates" column of the Eastern Edition of The Wall Street Journal on the Closing Date.

- (74) **"Purchase Price**" shall have the meaning set forth in Section 2.2.
- (75) **"Real Property**" has the meaning set forth in Section 4.9.
- (76) **"Reference Net Working Capital"** means \$-3,550,054.

(77) **"Representatives**" means with respect to a particular Person, any agent, consultant, advisor, accountant, financial advisor, legal counsel or other representative of that Person.

- (78) **"Rules**" shall have the meaning set forth in Section 10.5(a).
- (79) "Seller" shall have the meaning set forth in the preamble to this Agreement.
- (80) **"Seller Claim**" shall have the meaning set forth in Section 9.3(b).
- (81) "Seller Disclosure Schedule" means the disclosure schedule of Seller referred to in, and delivered pursuant to, this Agreement.

(82) **"Seller's Knowledge**" means the actual knowledge of Greg Galbraith and Julie Sartor, after reasonable inquiry of each of their respective direct reports.

(83) "Seller Required Regulatory Approvals" shall have the meaning set forth in Section 4.3(b).

(84) "**Software**" means any and all (i) computer programs, including any and all software implementations of algorithms, models and methodologies, whether in source code or object code, (ii) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (iii) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons, and (iv) all documentation, including user manuals and other training documentation, related to any of the foregoing.

(85) **"Subsidiary**" of any Person (the "Subject Person") means any Person, whether incorporated or unincorporated, of which (i) at least 50% of the securities or ownership interests having by their terms ordinary voting power to elect a majority of the board of directors or other Persons performing similar functions, (ii) a general partner interest or (iii) a managing member interest, is

9

directly or indirectly owned or controlled by the Subject Person or by one or more of its respective Subsidiaries.

(86) "Survival Period" shall have the meaning set forth in Section 9.1(c).

(87) **"Tax**" means any tax, charge, fee, levy, penalty or other assessment imposed by any U.S. federal, state, local or foreign Taxing Authority, including any excise, property, income, sales, transfer, franchise, payroll, withholding, social security or other tax, including any interest, penalties or additions attributable thereto.

(88) **"Tax Return**" means any return, report, information return, declaration, claim for refund or other document (including any related or supporting information) supplied or required to be supplied to any authority with respect to Taxes and including any supplement or amendment thereof.

(89) "**Taxing Authority**" means the Internal Revenue Service and any other Governmental Authority responsible for the administration

of any Tax.

(90) **"Termination Date**" shall have the meaning set forth in Section 8.1(b).

(91) "TCPL" means TCPL Tuscarora Ltd., a Delaware corporation.

(92) "Third Party Claim" shall have the meaning set forth in Section 9.4(a).

(93) **"Transfer Taxes**" means any and all transfer Taxes, including sales, use, excise, stock, stamp, documentary, filing, recording, permit, license, authorization and similar Taxes, fees, duties, levies, customs, tariffs, imposts, assessments, obligations and charges.

#### ARTICLE II

#### PURCHASE AND SALE OF GENERAL PARTNERSHIP INTEREST

2.1 <u>Purchase and Sale</u>. Upon the terms and subject to the satisfaction (or waiver, if permitted) of the conditions contained in this Agreement, Buyer agrees to purchase from Seller and Seller agrees to sell, assign, convey, transfer and deliver to Buyer, the Interest.

2.2 <u>Purchase Price</u>. Pursuant to Article III, at the Closing, Buyer shall pay, in consideration for the purchase of the Interest pursuant to Section 2.1, in cash the sum of (a) \$98,000,000 and (b) 49% of the Estimated Cash Amount (together, the "**Initial Purchase Price**"), as adjusted by the Estimated Closing Adjustment pursuant to Section 2.3(b) (the "**Closing Consideration**"). The Closing Consideration is subject to

10

adjustment following the Closing by the Final Closing Adjustment and, as so adjusted, is referred to herein as the "Purchase Price."

#### 2.3 <u>Purchase Price Adjustment</u>.

(a) No later than two (2) Business Days prior to the Closing Date, Seller shall prepare and deliver to Buyer a certificate of an officer of Seller setting forth its good faith estimate of (i) the aggregate amount of cash and cash equivalents (including money market funds) of the Partnership as of the Closing Date (the "Estimated Cash Amount"), (ii) the Net Working Capital as of the Closing Date (the "Estimated Net Working Capital"), (iii) the aggregate amount of any capital expenditures not contemplated by the 2006 or 2007 capital expenditure budgets of the Partnership that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance with Section 6.1 (the "Estimated CapEx Amount") and (iv) the aggregate amount of any principal and interest payments with respect to indebtedness of the Partnership that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance **Weak Beattership** that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance **Weak Beattership** that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance with **Section 6.1** (the "Estimated Debt Repayment Amount").

(b) The Initial Purchase Price shall be (i) (A) increased, if the Estimated Net Working Capital exceeds the Reference Net Working Capital, by an amount equal to 49% of the amount of such excess, or (B) decreased, if the Reference Net Working Capital exceeds the Estimated Net Working Capital, by an amount equal to 49% of the amount of such excess, (ii) increased by an amount equal to 49% of any Estimated Debt Repayment Amount (the aggregate of such increase(s) or decrease, as the case may be, being the "Estimated Closing Adjustment").

(c) Within 45 days following the Closing Date, Seller shall prepare and deliver to Buyer, together with any supporting documentation, the following (collectively, the **"Preliminary Statement"**):

(i) an unaudited balance sheet of the Partnership as of the Closing Date (before giving effect to the Closing) (the **"Preliminary Closing Balance Sheet"**), prepared by Seller in accordance with GAAP, consistently applied;

(ii) a calculation by Seller of the aggregate amount of cash and cash equivalents (including money market funds) of the Partnership as of the Closing Date (before giving effect to the Closing) based on the Preliminary Closing Balance Sheet (the "**Preliminary Cash Amount**");

(iii) a calculation by Seller of the Net Working Capital as of the Closing Date (before giving effect to the Closing) based on the Preliminary Closing Balance Sheet (the "**Preliminary Net Working Capital**");

11

(iv) a calculation by Seller of the aggregate amount of any capital expenditures not contemplated by the 2006 or 2007 capital expenditure budgets of the Partnership that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance with Section 6.1 (the "**Preliminary CapEx Amount**"); and

(v) a calculation by Seller of the aggregate amount of any principal and interest payments with respect to indebtedness of the Partnership that may be made by the Partnership after the date of this Agreement and prior to the Closing in accordance with Section 6.1 (the **"Preliminary Debt Repayment Amount"**).

(d) In connection with the preparation of the Preliminary Statement, Seller and its representatives shall have reasonable access, during normal business hours and upon reasonable notice, to the books and records, the financial systems and finance personnel and any other information of the Partnership that Seller reasonably requests, and Buyer shall, and shall use its commercially reasonable efforts to cause the Partnership, to cooperate reasonably with Seller and its representatives in connection therewith.

(e) Buyer shall have fifteen (15) Business Days following receipt of the Preliminary Statement to review the Preliminary Closing Balance Sheet and the calculations of the Preliminary Cash Amount, the Preliminary Net Working Capital, the Preliminary CapEx Amount and the Preliminary Debt Repayment Amount, and to notify Seller in writing if it disputes the amount of the Preliminary Cash Amount, the Preliminary Net Working Capital, the Preliminary CapEx Amount or the Preliminary Debt Repayment Amount set forth on the Preliminary Statement (the "**Dispute Notice**"), specifying the reasons therefor in reasonable detail (and providing any supporting documentation).

(f) In connection with Buyer's review (and subject to the Confidentiality Agreement), Buyer and its Representatives shall have reasonable access, during normal business hours and upon reasonable notice, to all relevant work papers, schedules, memoranda and other documents prepared by Seller or its Representatives (subject to customary indemnification agreements with respect to work papers of the Partnership's or Seller's independent accountants that may be requested by such independent accountants) in connection with Seller's preparation of the Preliminary Closing Balance Sheet and/or its calculation of the Preliminary Cash Amount, the Preliminary Net Working Capital, the Preliminary CapEx Amount or the Preliminary Debt Repayment Amount and to finance personnel of Seller and any other information which Buyer reasonably requests, and Seller shall cooperate reasonably with Buyer and its Representatives in connection therewith.

(g) In the event that Buyer shall deliver a Dispute Notice to Seller, Buyer and Seller shall cooperate in good faith to resolve such dispute as promptly as practicable and, upon such resolution, if any, any adjustments to the Preliminary Closing Balance Sheet, the Preliminary Cash Amount, the Preliminary Net Working Capital, the

12

Preliminary CapEx Amount and the Preliminary Debt Repayment Amount shall be made in accordance with the agreement of Buyer and Seller. In connection with Seller's review of the Dispute Notice, Seller and its Representatives shall have reasonable access, during normal business hours and upon reasonable notice, to all relevant work papers, schedules, memoranda and other documents prepared by Buyer or its Representatives (subject to customary indemnification agreements with respect to work papers of Buyer's independent accountants that may be requested by such independent accountants) in connection with Buyer's preparation of the Dispute Notice and to finance personnel of Buyer and any other information which Seller reasonably requests, and Buyer shall cooperate reasonably with Seller and its Representatives in connection therewith. If Buyer and Seller are unable to resolve any such dispute within fifteen (15) Business Days (or such longer period as Buyer and Seller shall mutually agree in writing) of Buyer's delivery of such Dispute Notice, such dispute shall be resolved by the Independent Accounting Firm, and such determination shall be final and binding on the parties. Any expenses relating to the engagement of the Independent Accounting Firm in respect of its services pursuant to this Section 2.3(g) shall be shared equally by Seller and Buyer. The Independent Accounting Firm shall be instructed to use commercially reasonable efforts to perform its services within thirty (30) days of submission of the Preliminary Closing Balance Sheet, the Preliminary Cash Amount, the Preliminary Net Working Capital, the Preliminary CapEx Amount and the Preliminary Debt Repayment Amount to it and, in any case, as promptly as practicable after such submission. The Preliminary Closing Balance Sheet, the Preliminary Cash Amount, the Preliminary Net Working Capital, the Preliminary CapEx Amount and the Preliminary Debt Repayment Amount, (i) if no Dispute Notice has been timely delivered by Buyer, as originally submitted by Seller, or (ii) if a Dispute Notice has been timely delivered by Buyer, as adjusted pursuant to the resolution of such dispute in accordance with this Section 2.3(g), shall be, respectively, the "Final Closing Balance Sheet," the "Final Cash Amount," the "Final Net Working Capital," the "Final CapEx Amount" and the "Final Debt Repayment Amount".

(h) The "**Purchase Price**" shall be equal to the Closing Consideration, adjusted as follows (the aggregate of such increases or decreases, as the case may be, pursuant to clauses (i) and (ii) below is referred to as the "**Final Closing Adjustment**"):

(i) (A) increased, if the Final Cash Amount exceeds the Estimated Cash Amount, by an amount equal to 49% of the amount of such excess, or (B) decreased, if the Estimated Cash Amount exceeds the Final Cash Amount, by an amount equal to 49% of the amount of such excess;

(ii) (A) increased, if the Final Net Working Capital exceeds the Estimated Net Working Capital, by an amount equal to 49% of the amount of such excess, or (B) decreased, if the Estimated Net Working Capital exceeds the Final Net Working Capital, by an amount equal to 49% of the amount of such excess;

amount

(iii) (A) increased, if the Final CapEx Amount exceeds the Estimated CapEx Amount, by an amount equal to 49% of the

13

of such excess, or (B) decreased, if the Estimated CapEx Amount exceeds the Final CapEx Amount, by an amount equal to 49% of the amount of such excess; and

(iv) (A) increased, if the Final Debt Repayment Amount exceeds the Estimated Debt Repayment Amount, by an amount equal to 49% of the amount of such excess, or (B) decreased, if the Estimated Debt Repayment Amount exceeds the Final Debt Repayment Amount, by an amount equal to 49% of the amount of such excess.

(i) Buyer or Seller, as the case may be, shall, within five (5) Business Days after the determination of the Final Cash Amount, the Final Net Working Capital, the Final CapEx Amount and the Final Debt Repayment Amount pursuant to this Section 2.3, make payment to the other by wire transfer in immediately available funds of the amount payable by Buyer or Seller, as the case may be, in an amount equal to the Final Closing Adjustment, together with interest thereon from the Closing Date to the date of payment at a rate equal to the Prime Rate (less any non-resident withholding Tax payable in respect of such interest). Such interest shall be calculated based on a year of 365 days and the number of days elapsed since the Closing Date.

#### ARTICLE III THE CLOSING

3.1 <u>Time and Place of Closing</u>. Upon the terms and subject to the satisfaction (or waiver, if permitted) of the conditions contained in this Agreement, the closing of the transactions contemplated by this Agreement (the "**Closing**") shall take place at the offices of Skadden, Arps, Slate, Meagher & Flom LLP, 4 Times Square, New York, New York, at 10:00 a.m., local time, no later than the third Business Day following the date on which all of the conditions to each party's obligations hereunder have been satisfied or waived (other than those conditions that by their nature have to be satisfied at the Closing (but subject to the satisfaction or waiver (if permitted) of those conditions)), or at such other place or time as the parties may agree. The date and time at which the Closing actually occurs is hereinafter referred to as the "**Closing Date**."

3.2 <u>Deliveries by Seller</u>. At the Closing, Seller shall deliver to Buyer the following:

(a) Certificates of the appropriate public officials as of a recent date to the effect that the Partnership is a validly existing Nevada general partnership in good standing in Nevada and each of the respective states in which it is registered to conduct business as a foreign entity;

(b) Incumbency and specimen signature certificates dated the Closing Date, signed by an appropriate officer of Seller and certified by an appropriate officer;

(c) An assignment and assumption agreement duly executed by Seller in respect of the Interest;

14

(d) The certificate referred to in Section 7.2(c);

(e) An affidavit of non-foreign status from Seller that complies with Section 1445 of the Code;

(f) A Transition Services Agreement, executed by Tuscarora Gas Operating Company and the Partnership, in form and substance reasonably acceptable to Seller and Buyer; and

(g) Such other agreements, documents, instruments and writings as are required to be delivered by Seller at or prior to the Closing Date pursuant to this Agreement or otherwise required in connection herewith.

3.3 <u>Deliveries by Buyer</u>. At the Closing, Buyer shall deliver to Seller the following:

(a) Incumbency and specimen signature certificates dated the Closing Date, signed by an appropriate officer of Buyer and certified by an appropriate officer;

(b) An assignment and assumption agreement duly executed by Buyer in respect of the Interest;

(c) The Closing Consideration by wire transfer of immediately available funds or by such other means as may be mutually agreed to

by Seller and Buyer;

(d) The certificate referred to in Section 7.3(c); and

(e) Such other agreements, documents, instruments and writings as are required to be delivered by Buyer at or prior to the Closing Date pursuant to this Agreement or otherwise required in connection herewith.

## ARTICLE IV

#### **REPRESENTATIONS AND WARRANTIES OF SELLER**

Except as set forth in the Seller Disclosure Schedule, Seller hereby represents and warrants to Buyer, as of the date of this Agreement, as

4.1 <u>Organization; Qualification</u>. The Partnership is a general partnership duly organized, validly existing and in good standing under the laws of the State of Nevada and has all requisite general partnership power and authority to own, lease, and operate its properties and to carry on the Business as is now being conducted. The Partnership is duly qualified or licensed to do business as a foreign general partnership and is in good standing in each jurisdiction set forth in <u>Section 4.1 of the Seller Disclosure Schedule</u>, which are the only jurisdictions in which the property owned, leased or operated by it or the nature of the Business conducted by it makes such qualification necessary, except where the failure to be so duly qualified or licensed and in good standing would not have or be reasonably expected to have, individually or in the

15

aggregate, a Material Adverse Effect. A complete and correct copy of the Partnership Agreement as currently in effect has been made available to Buyer.

4.2 <u>Authority Relative to this Agreement and Related Agreements</u>. Seller has the corporate power and authority to execute and deliver this Agreement and the Interest Holder Agreement and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Seller of this Agreement and the Interest Holder Agreement and the consummation by Seller of the transactions contemplated hereby and thereby have been duly and validly authorized by Seller and no other corporate proceedings on the part of Seller are necessary to authorize this Agreement have been duly executed and delivered by Seller and, assuming that this Agreement and the Interest Holder Agreement constitute legal, valid and binding agreements of Buyer, constitute legal, valid and binding agreements of Seller, and are enforceable against Seller in accordance with their terms, except that such enforceability may be limited by applicable bankruptcy, insolvency, moratorium or other similar laws affecting or relating to enforcement of creditors' rights generally or general principles of equity.

#### 4.3 Consents and Approvals; No Violation

follows:

(a) Except as set forth in Section 4.3(a) of the Seller Disclosure Schedule, and other than obtaining the Seller Required Regulatory Approvals and the Buyer Required Regulatory Approvals, the execution, delivery and performance of this Agreement and the Interest Holder Agreement by Seller and the consummation by Seller of the transactions contemplated hereby and thereby will not (i) conflict with or result in any breach of any provision of the articles of incorporation or bylaws of Seller, (ii) result in a violation or default (with or without notice or lapse of time, or both) or give rise to any right of termination, cancellation or acceleration of any obligation or to loss of a benefit under any of the terms, conditions or provisions of any Permit, Contract or other instrument or obligation to which Seller or the Partnership is a party or by which their respective assets may be bound or (iii) violate or breach any law, order, injunction, decree, statute, rule or regulation applicable to Seller or the Partnership, except, in the case of clauses (ii) and (iii) above, for violations, defaults, or rights of termination, cancellation or acceleration, losses of benefits or breaches which would not have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect.

(b) Except as set forth in <u>Section 4.3(b) of the Seller Disclosure Schedule</u> and except for the filings by Seller and Buyer required by the HSR Act and the expiration or earlier termination of all waiting periods under the HSR Act (the filings and approvals referred to in <u>Section 4.3(b) of the Seller Disclosure Schedule</u> and above are collectively referred to as the "**Seller Required Regulatory Approvals**"), no material declaration, filing or registration with, or notice to, or authorization, consent or approval of any Governmental Authority (other than the Permits and Environmental Permits) is necessary for the consummation by Seller of the transactions contemplated by this Agreement or the Interest Holder Agreement.

16

4.4 <u>Capitalization</u>. Section 4.4 of the Seller Disclosure Schedule sets forth all of the outstanding and authorized general partnership interests of the Partnership and all such interests are owned by the Persons as described in such Section of the Seller Disclosure Schedule. The Interest has been duly authorized and validly issued, and is fully paid and nonassessable, and is held of record and beneficially owned by Seller. Except as set forth in Section 4.4 of the Seller Disclosure Schedule or in the Partnership Agreement, there are no preemptive rights, rights of first refusal or any outstanding subscriptions, options, warrants or similar agreements of any kind to which Seller or the Partnership is party or by which the Partnership interests, or obligating it to issue, deliver or sell additional general partnership interests or any securities convertible or exchangeable into general partnership interests, or obligating the Partnership to grant, extend or enter into any such subscription, option, warrant or agreement. At the Closing, Seller will transfer to Buyer good title to the Interest, free and clear of Encumbrances (other than Encumbrances created by Buyer or arising under applicable securities laws). The Partnership does not own, directly or indirectly, any capital stock or equity securities of any Person.

4.5 <u>Financial Statements</u>. Seller has delivered or made available copies to Buyer of the Financial Statements. The Financial Statements were prepared in accordance with GAAP, consistently applied (except as disclosed in the notes thereto), and fairly present, in all material respects, the financial position and results of operations of the Partnership as of the dates thereof and for the periods covered thereby.

4.6 <u>Undisclosed Liabilities</u>. Except as set forth in <u>Section 4.6 of the Seller Disclosure Schedule</u>, the Partnership has no material liability relating to the Business of a nature required by GAAP to be reflected in a balance sheet of the Partnership or the notes thereto) except for those liabilities (a) accrued or reserved against in the Interim Balance Sheet or disclosed in the notes thereto, (b) incurred in the ordinary course of business consistent with past practice after the Balance Sheet Date or (c) incurred in accordance with Section 6.1 after the date of this Agreement.

4.7 <u>Absence of Certain Changes or Events</u>. Except as set forth in <u>Section 4.7 of the Seller Disclosure Schedule</u>, since the Balance Sheet Date to the date of this Agreement, (a) there has not been any event, change, occurrence or circumstances that has had, or would reasonably be expected to have, a Material Adverse Effect, (b) the Partnership has conducted the Business only in the ordinary course consistent with past practice and (c) the Partnership has not suffered any loss, damage, destruction or other casualty to any of its property, plant, equipment or inventories (whether or not covered by insurance) in excess of \$500,000.

4.8 <u>Compliance with Law</u>. Except for Environmental, Permits and Tax matters, which are the subject of Section 4.13, Section 4.17 and Section 4.18, respectively, and except as set forth in <u>Section 4.8 of the Seller Disclosure Schedule</u>, since January 1, 2005, each of Seller (solely with respect to the Business) and the Partnership has complied with all laws, statutes, ordinances, rules and regulations of any Governmental Authority applicable to its properties, assets and Business, except where such noncompliance would not have or be reasonably expected to have, individually or in

the aggregate, a Material Adverse Effect. Since January 1, 2005, neither Seller (solely with respect to the Business) nor the Partnership has received notice of any material violation of any such law, statute, ordinance, rule or regulation. To Seller's Knowledge, neither Seller (solely with respect to the Business) nor the Partnership is under any investigation by any Governmental Authority with respect to the material violation of any such laws, statutes, ordinances, rules and regulations of any Governmental Authority.

4.9 <u>Title to Real Property</u>. Except as set forth in <u>Section 4.9 of the Seller Disclosure Schedule</u>, the Partnership has valid and indefeasible title to or holds a valid, binding and enforceable leasehold, license or other interest in, or right-of-way or easement through, all real property used by the Partnership in the ordinary course of business consistent with past practice (the "**Real Property**"), in each case free and clear of all Encumbrances.

4.10 <u>Intellectual Property</u>. <u>Section 4.10 of the Seller Disclosure Schedule</u> sets forth an accurate and complete list of all material registrations or applications for registration for United States copyrights, patents, or trademarks or servicemarks (including Internet domain names) owned by the Partnership, including the jurisdiction and record owner of each application or registration. Except as would not have or reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect, or as set forth on <u>Section 4.11 of the Seller Disclosure Schedule</u>, to Seller's Knowledge, (a) the Partnership owns or has valid rights to use all Intellectual Property necessary to operate the Business as currently conducted and (b) there are no pending or threatened written claims that (i) the Partnership is in violation of, infringing upon, diluting or misappropriating any Intellectual Property rights of any third Person or (ii) challenge the validity, enforceability or ownership of any material Intellectual Property owned by the Partnership. To Seller's Knowledge, no third Person is infringing, violating, diluting, or misappropriating any of the material Intellectual Property owned by the Partnership.</u>

4.11 <u>Material Leases. Section 4.11 of the Seller Disclosure Schedule</u> sets forth a list of all Material Leases, including the name of the third party and the date of the lease or sublease and all amendments thereto, as of the date of this Agreement. Except as set forth in <u>Section 4.11 of the Seller</u> <u>Disclosure Schedule</u>, (a) each Material Lease is valid, binding and enforceable in accordance with its terms, and is in full force and effect, (b) the Partnership has not received any notice of material default of the Partnership under any Material Lease in the 12-month period prior to the date of this Agreement and (c) to the Seller's Knowledge, there are no uncured defaults of the Partnership under any Material Lease that would give the counterparty thereof the right to terminate such Material Lease.

4.12 <u>Insurance</u>. Section 4.12 of the Seller Disclosure Schedule sets forth a list of all insurance policies and all fidelity bonds held by or applicable to Seller (solely with respect to the Business) and the Partnership setting forth, in respect of each such policy, the policy name, policy number, carrier, term, type and amount of coverage and annual premium. Except as set forth in <u>Section 4.12 of the Seller Disclosure Schedule</u>, all material policies of fire, liability, workers' compensation and other forms of

18

insurance purchased or held by and insuring the Partnership or the Business are in full force and effect, all premiums with respect thereto have been paid, and no written notice of cancellation or termination has been received with respect to any such policy which was not replaced on substantially similar terms prior to the date of such cancellation.

4.13 <u>Environmental Matters</u>. (a) Except as set forth in <u>Section 4.13(a) of the Seller Disclosure Schedule</u>, the Partnership holds, and is in substantial compliance with, all material permits, licenses and governmental authorizations (the "**Environmental Permits**") required for the Partnership to operate the Business under applicable Environmental Laws, and the Partnership is otherwise in compliance with applicable Environmental Laws with respect to the Business except for such failures to hold or comply with required Environmental Permits, or such failures to be in compliance with applicable Environmental Laws, which would not have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect.

(b) Except as set forth in <u>Section 4.13(b) of the Seller Disclosure Schedule</u>, the Partnership has not received any written request for information, or been notified in writing that it is a potentially responsible party, under CERCLA or any similar state law with respect to the Business.

(c) Except as set forth in <u>Section 4.13(c) of the Seller Disclosure Schedule</u>, the Partnership has not entered into or agreed to any material consent, decree or order, and is not subject to any judgment, decree, or judicial order relating to compliance with any Environmental Law or to investigation or cleanup of Hazardous Substances under any Environmental Law which is outstanding and imposes material obligations on the Partnership.

(d) To Seller's Knowledge, no facts, circumstances or conditions exist that would reasonably be expected to result in liability under any Environmental Law having a Material Adverse Effect.

(e) Seller has delivered or made available to Buyer copies of all reports, assessments, investigations, or correspondence related to the Partnership in each case that relate to environmental matters and are material to the extent in the possession, custody, or control of Seller.

(f) Save and except those representations and warranties in Section 4.3, the representations and warranties made in this Section 4.13 are Seller's exclusive representations and warranties relating to environmental matters.

4.14 <u>Employee Matters</u>. There are no employees of the Partnership and no employee or employee-benefit related liabilities of the Partnership.

4.15 <u>Material Contracts</u>. (a) <u>Section 4.16(a) of the Seller Disclosure Schedule</u> sets forth a list of all Contracts to which the Seller (solely with respect to the Business) or the Partnership, as applicable, is a party as of the date of this Agreement (excluding, for the avoidance of doubt, any such Contracts that are no longer in effect):

(i) for joint ventures, strategic alliances, partnerships, material licensing agreements (excluding licenses of generally available or commercial Software), or sharing of profits;

(ii) containing (i) covenants of the Partnership not to compete with any Person in any line of business or in any geographical area or (ii) covenants of any other Person not to compete with the Partnership in any line of business or in any geographical area;

(iii) with respect to all price swaps, hedges, futures or similar instruments;

(iv) with any current or former officer, director, stockholder, member, partner (or immediate family member thereof) of Seller, any Seller Affiliate or the Partnership;

(v) for the sale of any of the material assets of the Partnership other than in the ordinary course of business consistent with past practice or for the grant to any Person of any preferential rights to purchase any of its material assets;

(vi) for the acquisition (by merger, purchase of stock or assets or otherwise) by the Partnership of any operating business or the capital stock of any other Person;

(vii) for the incurrence, assumption or guarantee of any indebtedness for borrowed money of the Partnership or imposing an Encumbrance on any of the assets of the Partnership in connection therewith, including indentures, guarantees, loan or credit agreements, sale and leaseback agreements, purchase money obligations incurred in connection with the acquisition of property, mortgages, pledge agreements, security agreements, or conditional sale or title retention agreements;

(viii) providing for payments by or to the Partnership in excess of \$250,000 in any fiscal year or \$500,000 in the aggregate during the term thereof; provided that the calculation of the aggregate payments for any such Contract shall not include payments attributable to any renewal periods or extensions for which the Partnership may exercise an option in its sole discretion to approve or disapprove;

(ix) under which the Partnership has made advances or loans to any other Person;

(x) providing for severance, retention, change in control or other similar payments; and

(xi) providing for the guaranty, surety or indemnification, direct or indirect, by the Partnership, which reasonably

20

could be expected to result in liability to the Partnership in excess of \$500,000 (each document set forth on Section 4.16-A of the Seller Disclosure Schedule, being a "Material Contract").

(b) Except as disclosed in <u>Section 4.15 of the Seller Disclosure Schedule</u>, (a) each Material Contract is valid, binding and enforceable in accordance with its terms, and is in full force and effect, (b) neither Seller nor the Partnership has received or is aware of any notice of its default under any Material Contract in the 12-month period prior to the date of this Agreement, (c) there are no uncured defaults of Seller or the Partnership under any Material Contract that would give the counterparty thereof the right to terminate such Material Contract, (d) no party to any of the Material Contracts has exercised any termination rights with respect thereto and (e) no party has given written notice of any significant dispute with respect to any Material Contract. Seller has delivered or made available to Buyer correct and complete copies of all of the Material Contracts, together with all amendments, modifications or supplements thereto. Neither Seller (solely with respect to the Business) nor the Partnership is a party to any material oral contract, agreement or arrangement.

4.16 Legal Proceedings, etc. Except as set forth in Section 4.16 of the Seller Disclosure Schedule, there are no material claims, actions, or proceedings pending or, to Seller's Knowledge, threatened against Seller or the Partnership relating to the Business, or any of their respective officers, directors or employees with respect to their business activities on behalf of the Partnership, by or before any Governmental Authority, including any claims, actions or proceedings relating to this Agreement or the transactions contemplated hereby, which, if adversely determined, would have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect. Except as set forth in Section 4.16 of the Seller Disclosure Schedule, neither Seller (solely with respect to the Business) nor the Partnership is (a) subject to any outstanding judgment, rule, order, writ, injunction or decree of any Governmental Authority relating to the Business which would have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect or (b) engaged in any legal action to recover monies due it or for damages sustained by it.

4.17 <u>Permits</u>. The Partnership has all permits, licenses, franchises and other governmental authorizations, consents and approvals, other than with respect to Environmental Laws, necessary to operate the Business as presently operated (collectively, "**Permits**"), except where the failure to have such Permits would not have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect. Except as set forth in <u>Section</u>

<u>4.17(b) of the Seller Disclosure Schedule</u>, (a) neither Seller (solely with respect to the Business) nor the Partnership has received any notification that it is in material violation of any Permits, (b) the Partnership is in compliance with all Permits and (c) all material Permits are in good standing, have not expired, are final and non-appealable and, to Seller's Knowledge, are not subject to any threat of revocation, amendment or modification, except, in the case of clause (b) and (c), as would not have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect.

Taxes. (a) To Seller's Knowledge, (i) the Partnership has filed all material Tax Returns required to be filed; (ii) such Tax Returns 4.18 are true, correct and complete in all material respects; and (iii) the Partnership has fully and timely paid all Taxes shown to be due on such Tax Returns. With respect to any period for which Tax Returns have not yet been filed or for which Taxes are not yet due or owing, the Partnership has made due and sufficient accruals for such Taxes in the Financial Statements and its books and records. To Seller's Knowledge, no claim has been made in writing by a Taxing Authority in a jurisdiction in which the Partnership does not currently file a Tax Return that the Partnership is or may be subject to taxation by that jurisdiction. Neither Seller nor, to Seller's Knowledge, the Partnership has received any written notice of deficiency or assessment from any Taxing Authority with respect to (i) liabilities for material Taxes of the Partnership, which have not been fully and timely paid or finally settled, unless such liabilities are being diligently contested in good faith through appropriate proceedings and provided adequate reserves have been established in accordance with GAAP in the Financial Statements of the Partnership, or (ii) adjustments to taxable income of the Partnership which have not been finally resolved. To Seller's Knowledge, no Tax Return of the Partnership is currently being audited or examined by any Taxing Authority, and neither Seller nor, to Seller's Knowledge, the Partnership has received any written notice that any Taxing Authority intends to conduct an audit or investigation with respect to any such Tax Return. There are no outstanding (i) agreements or waivers extending the applicable statutory periods of limitation for Taxes of the Partnership, (ii) extensions for the assessment or collection of Taxes of the Partnership, which Taxes have not since been paid, or (iii) powers of attorney that are currently in force with respect to any Tax matter with respect to the Partnership, for any period. The Partnership has qualified as, and has been treated as, a partnership for U.S. federal income Tax purposes at all times since the date of its formation.

(b) The Partnership has complied in all material respects with all applicable laws relating to the payment and withholding of Taxes and has duly and timely withheld and paid over to the appropriate Taxing Authority all amounts required to be so withheld and paid under all applicable laws.

(c) Neither Seller nor, to Seller's Knowledge, the Partnership has executed or entered into any written agreement with, or obtained or applied for any written consents or written clearances or any other Tax rulings from, nor has there been any written agreement executed or entered into on behalf of either of them with, any Taxing Authority, including any IRS closing agreements pursuant to Section 7121 of the Code or any similar provision of law, private letter rulings or comparable rulings of any Taxing Authority.

(d) Seller is not a foreign person within the meaning of Section 1445 of the Code.

4.19 <u>Brokers' Fee</u>. Except as set forth in <u>Section 4.19 of the Seller Disclosure Schedule</u>, no broker, finder or similar intermediary has acted for or on behalf of, or is entitled to any broker, finder or similar fee or other commission from Seller or

22

any of its Affiliates in connection with this Agreement or the transactions contemplated hereby.

## ARTICLE V <u>REPRESENTATIONS AND WARRANTIES OF BUYER</u>

Except as set forth in the Buyer Disclosure Schedule, Buyer hereby represents and warrants to Seller, as of the date of this Agreement, as follows:

5.1 <u>Organization</u>. Buyer is a limited partnership duly organized, validly existing and in good standing under the laws of the State of Delaware and has all requisite limited partnership power and authority to own, lease and operate its properties and to carry on its business as now being conducted. Complete and correct copies of Buyer's certificate of limited partnership and partnership agreement (or other similar governing documents) as currently in effect have heretofore been delivered or made available to Seller.

5.2 <u>Authority Relative to this Agreement</u>. Buyer has the limited partnership power and authority to execute and deliver this Agreement and the Interest Holder Agreement and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance by Buyer of this Agreement and the Interest Holder Agreement and the consummation by Buyer of the transactions contemplated hereby and thereby and thereby have been duly and validly authorized by Buyer and no other limited partnership proceedings on the part of Buyer are necessary to authorize this Agreement or Interest Holder Agreement or to consummate the transactions contemplated hereby or thereby. This Agreement and the Interest Holder Agreement have been duly executed and delivered by Buyer, and assuming that this Agreement and the Interest Holder Agreement constitute legal, valid and binding agreements of Seller, constitute legal, valid and binding agreements of Buyer, and are enforceable against Buyer in accordance with their terms, except that such enforceability may be limited by applicable bankruptcy, insolvency, moratorium or other similar laws affecting or relating to enforcement of creditors' rights generally or general principles of equity.

5.3 <u>Consents and Approvals; No Violation</u>. (a) Except as set forth in <u>Section 5.3(a) of the Buyer Disclosure Schedule</u>, and other than obtaining the Buyer Required Regulatory Approvals and the Seller Required Regulatory Approvals, the execution, delivery and performance of this Agreement and the Interest Holder Agreement by Buyer and the consummation by Buyer of the transactions contemplated hereby and thereby will not (i) conflict with or result in any breach of any provision of the certificate of limited partnership or partnership agreement (or similar organizational documents) of Buyer, (ii) result in a violation or default (with or without notice or lapse of time, or both) or give rise to any right of termination, cancellation or acceleration of any obligation or to loss of a benefit under any of the terms, conditions or provisions of any Permit, Contract, or other instrument or obligation to which Buyer is a party or by which its assets may be bound or (ii) violate or breach any law, order, injunction, decree, statute, rule or regulation applicable to Buyer, except, in the case of clauses (ii) and (iii) above, for violations, defaults, or rights of termination, cancellation or acceleration,

losses of benefits or breaches which would not or would not be reasonably expected to, individually or in the aggregate, materially impair Buyer's ability to perform its obligations under this Agreement or the Interest Holder Agreement or consummate the transactions contemplated hereby or thereby.

(b) Except as set forth in <u>Section 5.3(b) of the Buyer Disclosure Schedule</u> and except for the filings by Buyer and Seller required by the HSR Act and the expiration or earlier termination of all waiting periods under the HSR Act (the filings and approvals referred to in <u>Section 5.3(b) of the Buyer Disclosure Schedule</u> and above are collectively referred to as the "**Buyer Required Regulatory Approvals**"), no material declaration, filing or registration with, or notice to, or authorization, consent or approval of any Governmental Authority is necessary for the consummation by Buyer of the transactions contemplated hereby.

5.4 <u>Availability of Funds</u>. On the Closing Date, Buyer will have immediately available funds sufficient to consummate the transactions contemplated by this Agreement.

5.5 <u>Acquisition for Investment; Due Diligence</u>. Buyer is an informed and sophisticated purchaser experienced in financial and business matters and the evaluation and investment in businesses such as the Business as contemplated hereunder. Buyer is acquiring the Interest for investment and not with a view toward or for sale in connection with any distribution thereof in violation of applicable securities laws. Buyer agrees that the Interest may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act of 1933, as amended, except pursuant to an exemption from such registration available under such Act. Buyer has undertaken such investigation and has been provided with and has evaluated such documents and information as it has deemed necessary to enable it, when considered together with the representations and warranties and covenants of Seller set forth herein, to make an informed and intelligent decision with respect to the execution, delivery and performance of this Agreement. Buyer has been furnished the materials relating to Buyer's purchase of the Interest that it has requested, and that Seller has provided Buyer the opportunity to ask questions of the officers and management employees of the Business and to acquire additional information about the Business and financial condition of the Business.

5.6 <u>Brokers' Fee</u>. No broker, finder or similar intermediary has acted for or on behalf of, or is entitled to any broker, finder or similar fee or other commission from Buyer or any of its Affiliates in connection with this Agreement or the transactions contemplated hereby.

## ARTICLE VI COVENANTS OF THE PARTIES

6.1 <u>Conduct of Business of the Partnership</u>. Except as set forth in <u>Section 6.1-A of the Seller Disclosure Schedule</u>, during the period from the date of this Agreement to the Closing Date, Seller shall (i) use its commercially reasonable efforts to

24

cause the Partnership to, and (ii) if applicable, vote all of its general partnership interests in the Partnership in a manner consistent with causing the Partnership to, operate the Business only in the ordinary course of business, consistent with past practice. Without limiting the generality of the foregoing, and, except as contemplated in this Agreement or as described in <u>Section 6.1-B of the Seller Disclosure Schedule</u>, prior to the Closing Date, without the prior written consent of Buyer (such consent not to be unreasonably withheld, delayed or conditioned), Seller shall (i) use its commercially reasonable efforts to cause the Partnership, and (ii) if applicable, vote all of its general partnership interests in the Partnership in a manner consistent with causing the Partnership, not to:

(a) (i) create, incur or assume any material amount of indebtedness for borrowed money, other than in the ordinary course of business consistent with past practice, including obligations in respect of capital leases but excluding purchase money mortgages granted in connection with the acquisition of property in the ordinary course of business, or (ii) assume, guarantee, endorse or otherwise become liable or responsible (whether directly, contingently or otherwise) for the obligations of any other Person except in the ordinary course of business consistent with past practice;

(b) make any material change in the operations of the Business;

(c) sell, lease (as lessor), transfer or otherwise dispose of, any of the material assets of the Business, other than in the ordinary course of business consistent with past practice, or mortgage or pledge, or impose or suffer to be imposed any Encumbrance on, any of the material assets of the Business;

(d) except as may be contemplated by the capital expenditure budget of the Partnership previously delivered or made available to Buyer, make any capital expenditures in excess of \$250,000 individually or \$500,000 in the aggregate;

(e) enter into or materially amend any Material Contracts or Permits or waive or release any material right thereunder other than in the ordinary course of business consistent with past practice;

(f) make any change in any material respect to any of the accounting principles, methods, policies or practices used by the Partnership, except for any change required by reason of a concurrent change in GAAP;

(g) make or change any material Tax elections or Tax accounting methods of the Partnership, except as required by applicable law, prepare or file any Tax Return (or amendment thereof) unless prepared in a manner consistent with past practice or settle or compromise any material Tax claim or liability;

(h) amend or change its organizational documents;

23

(i) (i) transfer, issue, sell, pledge, encumber, dispose or deliver any general partnership interests or any securities convertible or exchangeable into general partnership interests; (ii) grant any rights to purchase or otherwise acquire any general partnership interests or any securities convertible or exchangeable into general

partnership interests; or (iii) amend in any material respect any of the terms of any such general partnership interests or any securities convertible or exchangeable into general partnership interests outstanding as of the date hereof;

(j) (i) repurchase or otherwise acquire any general partnership interest of the Partnership; or (ii) adopt a plan of complete or partial liquidation or resolutions providing for or authorizing a liquidation, dissolution, consolidation, restructuring, recapitalization, or other reorganization of the Partnership;

(k) acquire (by merger, consolidation, or acquisition of stock or assets or otherwise) any corporation, partnership, or other business organization or division thereof;

(1) settle or compromise any pending or threatened legal proceeding or any claim or claims for, or that would result in Losses by the Partnership following the Closing of, an amount that could, individually or in the aggregate, reasonably be expected to be greater than \$500,000;

(m) (i) hire any employees prior to Closing; or (ii) establish, adopt or enter into any collective bargaining agreement or benefits plan, except as required under applicable law; or

(n) agree or otherwise commit to enter into any Contract, whether written or oral, with respect to any of the transactions set forth in the foregoing paragraphs (a) through (m).

6.2 Access to Information. (a) Between the date of this Agreement and the Closing Date, Seller shall (solely with respect to the Business), during ordinary business hours and upon reasonable notice, (i) give Buyer and its Representatives reasonable access to its personnel and all books, records, plants, offices and other facilities and properties constituting and solely related to the Business and the Partnership in Seller's possession and control which Buyer is permitted access by law as Buyer may reasonably request, (ii) permit Buyer to make such reasonable inspections thereof as Buyer may reasonably request; (iii) cause its officers and advisors to as promptly as reasonably practicable furnish Buyer with such financial and operating data and other information with respect to the Business in Seller's possession and control as Buyer may from time to time reasonably request; and (iv) cause its officers and advisors to as promptly as reasonably practicable furnish Buyer acopy of each report, schedule or other document filed or received by them with the FERC with respect to the Business in Seller's possession and control which Buyer is permitted access by law as Buyer may reasonably request; provided, however, that (A) any such investigation shall be conducted in such a manner as not to interfere unreasonably with the operation of the Business or the businesses of Seller or its Affiliates, (B) neither Seller nor the Partnership shall be required to take any action which would constitute a waiver of the attorney-client privilege and (C) neither Seller nor the Partnership shall be required to supply Buyer with any information which is subject to a reasonable legal obligation not to be supplied; provided, further, that in the case of (B) above, Seller will give reasonable access to such

26

information to Buyer upon the execution by Buyer and Seller of a joint defense agreement, the terms of which shall be mutually agreed upon by such parties. Notwithstanding anything in this Section 6.2 to the contrary, (i) Seller and the Partnership shall not be obligated to furnish or provide such access to medical records to the extent prohibited by applicable law and (ii) Buyer shall not have the right to perform or conduct any environmental sampling or testing at, in, on or underneath any of the property used in the operation of the Business.

(b) All information furnished to or obtained by Buyer and its Representatives pursuant to this Section 6.2 shall be subject to the provisions of the Confidentiality Agreement and shall be treated as confidential pursuant to such agreement.

(c) For a period of seven (7) years after the Closing Date, Seller and its Representatives shall have reasonable access to, including the right to photocopy, scan, fax or otherwise reproduce, all of the books and records of the Business in Buyer's possession and control to the extent that such access may reasonably be required by Seller in connection with matters relating to or affected by the operation of the Business prior to the Closing Date; provided, however, that any such actions shall be conducted in such a manner as not to interfere unreasonably with the operation of the Business. Such access shall be afforded by Buyer upon receipt of reasonable advance notice and during normal business hours. Seller shall be solely responsible for any costs or expenses incurred by them pursuant to this Section 6.2(c). If Buyer shall desire to dispose of any such books and records prior to the expiration of such seven-year period, Buyer shall, prior to such disposition, give Seller a reasonable opportunity, at Seller's expense, to segregate and remove such books and records as Seller may select.

6.3 <u>Expenses</u>. Except to the extent otherwise specifically provided herein, whether or not the transactions contemplated hereby are consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be borne by the party incurring such costs and expenses.

6.4 <u>Further Assurances</u>. Subject to the terms and conditions of this Agreement, each of the parties hereto shall use all commercially reasonable efforts to take, or cause to be taken, all action, and to do, or cause to be done, all things necessary, proper or advisable under applicable laws and regulations to consummate and make effective the sale of the Interest pursuant to this Agreement. From time to time after the date hereof, without further consideration, Seller shall, at Seller's expense, execute and deliver such documents to Buyer as Buyer may reasonably request in order more effectively to vest in Buyer ownership of the Interest. From time to time after the date hereof, Buyer shall, at Buyer's own expense, execute and deliver such documents to Seller as Seller may reasonably request in order to more effectively consummate the sale of the Interest pursuant to this Agreement. Notwithstanding anything to the contrary in this Section 6.4, no party shall be required to (a) expend funds or assume liabilities beyond those that are reasonable in nature and amount in the context of the transactions contemplated hereunder, (b) divest any of its assets, including its businesses, divisions or

properties or (c) agree to restrictions on its businesses, operations or conduct other than those that have been expressly agreed to in this Agreement.

6.5 <u>Public Statements</u>. After the execution of this Agreement, the parties shall consult with each other prior to issuing any public announcement, statement or other disclosure with respect to this Agreement, the Interest Holder Agreement or the transactions contemplated hereby or thereby, and neither party shall issue any such public announcement, statement or other disclosure without having first received the written consent of the other party (such consent not to be unreasonably withheld or delayed), except as may be required by law and except that the parties may make public announcements, statements or other disclosures with respect to this Agreement and the transactions contemplated hereby consistent with the terms and conditions of the Confidentiality Agreement.

6.6 <u>Consents and Approvals</u>. (a) Seller and Buyer shall each file or cause to be filed with the Federal Trade Commission and the United States Department of Justice any notifications required to be filed under the HSR Act and the rules and regulations promulgated thereunder with respect to the transactions contemplated hereby and by the Interest Holder Agreement as promptly as practicable after the date of this Agreement. The parties shall consult with each other as to the appropriate time of filing such notifications and shall use their best efforts to make such filings at the agreed upon time, to respond promptly to any requests for additional information made by either of such agencies, and to cause the waiting periods under the HSR Act to terminate or expire at the earliest possible date after the date of filing.

(b) Seller and Buyer shall cooperate with each other and (i) promptly prepare and file all necessary documentation, (ii) effect all necessary applications, notices, petitions and filings, and execute all agreements and documents, and (iii) use all commercially reasonable efforts to obtain all necessary consents, approvals and authorizations of all other Persons, in the case of each of the foregoing clauses (i), (ii), and (iii), necessary or advisable to consummate the transactions contemplated by this Agreement and the Interest Holder Agreement (including the Seller Required Regulatory Approvals) or required by the terms of any deed of trust, franchise, Permit, concession, Contract or other instrument to which Seller, the Partnership, or Buyer is a party or by which any of them is bound. Seller shall have the right to review and approve in advance all characterizations of the information relating to the Business, and each of Seller and Buyer shall have the right to review and approve in advance all characterizations of the information relating to the transactions contemplated by this Agreement, which appear in any filing made in connection with the transactions contemplated hereby. The parties agree that they shall consult with each other with respect to the transferring to Buyer or the obtaining by Buyer of all such necessary Environmental Permits, Permits, consents, approvals and authorizations of all third parties and Governmental Authorities. Seller and Buyer shall designate separate counsel with respect to all applications, notices, petitions and filings (joint or otherwise) relating to this Agreement and the transactions contemplated hereby on behalf of Seller, on the one hand, and Buyer, on the other hand, with all Governmental Authorities.

(c) The parties hereto shall consult with each other prior to proposing or entering into any stipulation or agreement with any Governmental Authority or any third party in connection with any federal, state or local governmental consents and approvals legally required for the consummation of the transactions contemplated hereby and shall not propose or enter into any such stipulation or agreement without the other party's prior written consent, which consent shall not be unreasonably withheld.

(d) Notwithstanding anything to contrary in this Section 6.4, no party shall be required to (i) expend funds or assume liabilities beyond those that are reasonable in nature and amount in the context of the transactions contemplated hereunder, (ii) divest any of its assets, including its businesses, divisions or properties or (iii) agree to restrictions on its businesses, operations or conduct other than those that have been expressly agreed to in this Agreement.

6.7 <u>Tax Matters</u>. (a) Notwithstanding any other provision of this Agreement, all Transfer Taxes incurred in connection with this Agreement and the transactions contemplated hereby shall be borne by Buyer, and Buyer shall, at its own expense, file, to the extent required by law, all necessary Tax Returns and other documentation with respect to all such Transfer Taxes, and, if required by applicable law, Seller shall join in the execution of any such Tax Returns or other documentation.

(b) Each of Buyer and Seller shall provide the other with such assistance as may reasonably be requested by the other party in connection with the preparation of any Tax Return, any audit or other examination by any Taxing Authority, or any judicial or administrative proceedings relating to liability for Taxes of the Partnership or with respect to the assets of the Partnership, and each shall retain and provide the requesting party with any records or information which may be relevant to such return, audit or examination, proceedings or determination. Any information obtained pursuant to this Section 6.7 or pursuant to any other Section hereof providing for the sharing of information or review of any Tax Return or other schedule relating to Taxes shall be kept confidential by the parties hereto.

6.8 <u>Supplements to Schedules</u>. On one or more occasions prior to the Closing Date, Seller may update, supplement or amend the Seller Disclosure Schedule with additional information to reflect events, facts, circumstances or occurrences arising (and, in the case of representation or warranties qualified by Seller's Knowledge, Seller's Knowledge obtained) after the date of this Agreement. Any such update, supplement or amendment shall be given effect and taken into account for purposes of this Agreement (including Article IX); <u>provided</u>, <u>however</u>, that any such update, supplement or amendment shall be disregarded and not taken into account for purposes of determining whether the conditions set forth in Section 7.2(b) or Section 7.2(d) have been satisfied.

6.9 <u>No Implied Representations</u>. Buyer hereby acknowledges and agrees that none of Seller, the Partnership, their Affiliates or any other Person has made or is making any representation or warranty whatsoever, express or implied, at law or in equity, except those representations and warranties of Seller explicitly set forth in Article IV of this Agreement. Without limiting the generality of the foregoing, except as

<sup>28</sup> 

explicitly set forth in Article IV of this Agreement, none of Seller, the Partnership, their Affiliates or any other Person has made or is making any representation, express or implied, with respect to Seller, the Partnership, their respective businesses or any of their assets, liabilities or operations, their past, current or future financial condition, profitability or performance, the value of any of their assets or any assets of the Business, the merchantability, suitability or fitness for a particular purpose or quality with respect to any of their assets or any assets of the Business or the condition or workmanship thereof or the absence of any defects therein (whether latent or patent), and any such other representations or warranties are hereby expressly disclaimed.

6.10 <u>Financial Statements</u>. During the period from the date hereof to the Closing, upon the reasonable request of Buyer from time to time, Seller shall as promptly as reasonably practicable provide Buyer with a copy of any financial statements and operating or management reports prepared by the Partnership in the ordinary course of business consistent with past practice that are in Seller's possession and control.

#### ARTICLE VII CLOSING CONDITIONS

7.1 <u>Conditions to Each Party's Obligations to Effect the Transactions Contemplated Hereby</u>. The respective obligations of each party to effect the transactions contemplated hereby shall be subject to the fulfillment or waiver (other than the conditions set forth in subsections (a) and (b) of this Section 7.1) at or prior to the Closing Date of the following conditions:

(a) No preliminary or permanent injunction or other order or decree by any federal or state court which prevents the consummation of the transactions contemplated hereby shall have been issued and remain in effect (each party agreeing to use its commercially reasonable efforts to have any such injunction, order or decree lifted) and no statute, rule or regulation shall have been enacted by any state or federal government or Governmental Authority in the United States which prohibits the consummation of the transactions contemplated hereby;

(b) The waiting period under the HSR Act applicable to the consummation of the transactions contemplated hereby shall have expired or been terminated;

(c) The Partnership Agreement shall be amended in form and substance reasonably acceptable to Seller and Buyer and duly executed by the requisite general partners; and

(d) (i) The Operating Agreement shall be terminated (without any liability to Seller, Tuscarora Gas Operating Company or their Affiliates) and (ii) the Partnership and Tuscarora Gas Operating Company shall have executed and delivered to each other customary mutual releases pursuant to which their respective obligations and liabilities under the Operating Agreement (other than any obligations and liabilities under

30

Section 5.01 and Section 5.02 thereof and other than any amounts due and owing under the Operating Agreement as of such termination) are forever released and discharged.

7.2 <u>Conditions to Obligations of Buyer</u>. The obligation of Buyer to effect the transactions contemplated by this Agreement shall be subject to the fulfillment at or prior to the Closing Date of the following additional conditions (any or all of which may be waived by Buyer in whole or in part in its sole discretion):

(a) Seller shall have performed and complied in all material respects with the covenants and agreements contained in this Agreement required to be performed and complied with by Seller on or prior to the Closing Date;

(b) The representations and warranties of Seller set forth in Article IV of this Agreement (disregarding any Material Adverse Effect or materiality qualifiers therein) shall be true and correct on and as of the Closing Date with the same force and effect as if made on and as of the Closing Date (except to the extent such representations and warranties shall have been expressly made as of an earlier date, in which case such representations and warranties shall have been expressly made as of an earlier date, in which case such representations and warranties shall have been true and correct only as of such earlier date), except for such failures to be true and correct which would not have or be reasonably expected to have, individually or in the aggregate, a Material Adverse Effect;

(c) Buyer shall have received a certificate from the authorized officer of Seller, dated the Closing Date, to the effect that to the best of such officer's knowledge, the conditions set forth in Sections 7.2(a) and 7.2(b) have been satisfied;

(d) Since the date of this Agreement, there has not been any event, change, occurrence or circumstances that has had, or would reasonably be expected to have, a Material Adverse Effect;

(e) The Partnership shall have in effect a valid election pursuant to Section 754 of the Code; and

(f) Seller shall have made all deliveries required under Section 3.2.

7.3 <u>Conditions to Obligations of Seller</u>. The obligation of Seller to effect the transactions contemplated by this Agreement shall be subject to the fulfillment at or prior to the Closing Date of the following additional conditions (any or all of which may be waived by Seller in whole or in part in its sole discretion):

(a) Buyer shall have performed in all material respects its covenants and agreements contained in this Agreement required to be performed and complied with by Buyer on or prior to the Closing Date;

(b) The representations and warranties of Buyer set forth in Article V of this Agreement (disregarding any materiality qualifiers therein) shall be true and correct on and as of the Closing Date with the same force and effect as if made on and as of the Closing Date (except to the extent such representations and warranties shall have been expressly made as of an earlier date, in which case such representations and

warranties shall have been true and correct only as of such earlier date), except for such failures to be true and correct which would not or be reasonably expected to, individually or in the aggregate, materially impair Buyer's ability to perform its obligations under this Agreement or the Interest Holder Agreement or consummate the transactions contemplated hereby or thereby;

(c) Seller shall have received a certificate from an authorized officer of Buyer, dated the Closing Date, to the effect that, to the best of such officer's knowledge, the conditions set forth in Sections 7.3(a) and 7.3(b) have been satisfied; and

(d) Buyer shall have made all deliveries required under Section 3.3.

## ARTICLE VIII TERMINATION

Seller.

8.1

Termination. (a) This Agreement may be terminated at any time prior to the Closing Date, by mutual written consent of Buyer and

(b) This Agreement may be terminated by Seller or Buyer if the transactions contemplated hereby shall not have been consummated on or before six (6) months from the date of this Agreement (the "**Termination Date**"); <u>provided</u>, <u>however</u>, that the right to terminate this Agreement under this Section 8.1(b) shall not be available to a party if its failure to fulfill any obligation under this Agreement has been the direct cause of, or directly resulted in, the failure of the Closing Date to occur on or before such date.

(c) This Agreement may be terminated by either Seller or Buyer if (i) any Governmental Authority, the consent of which is a condition to the obligations of Seller and Buyer to consummate the transactions contemplated hereby, shall have determined not to grant its consent and all appeals of such determination shall have been taken and have been unsuccessful, or (ii) any court of competent jurisdiction in the United States or any state shall have issued an order, judgment or decree permanently restraining, enjoining or otherwise prohibiting the transactions contemplated hereby and such order, judgment or decree shall have become final and nonappealable.

(d) This Agreement may be terminated by Buyer, if there has been a material violation or breach by Seller of any agreement, representation or warranty contained in this Agreement which has not been cured within thirty (30) days or which has rendered the satisfaction of any condition to the obligations of Buyer impossible and such violation or breach has not been waived by Buyer.

(e) This Agreement may be terminated by Seller, if there has been a material violation or breach by Buyer of any agreement, representation or warranty contained in this Agreement which has not been cured within thirty (30) days (except with respect to the obligations by Buyer to pay the Closing Consideration pursuant to Section 2.2 after all conditions to Buyer's obligations hereunder have been satisfied or waived, whereby Seller may immediately terminate this Agreement) or which has

32

rendered the satisfaction of any condition to the obligations of Seller impossible and such violation or breach has not been waived by Seller.

8.2 <u>Procedure and Effect of Termination</u>. In the event of termination of this Agreement by Buyer or Seller pursuant to Section 8.1, written notice thereof shall forthwith be given by the terminating party to the other party and this Agreement shall terminate and the transactions contemplated hereby shall be abandoned, without further action by any of the parties hereto. If this Agreement is terminated as provided herein:

(a) such termination shall be the sole remedy of the parties hereto with respect to breaches of any agreement, representation or warranty contained in this Agreement and none of the parties hereto nor any of their respective trustees, directors, officers, members, general partners, or Affiliates, as the case may be, shall have any liability or further obligation to the other party or any of their respective trustees, directors, officers, members, general partners, or Affiliates, as the case may be, pursuant to this Agreement, except in each case as specifically provided in this Section 8.2 and in Sections 6.2(b) and 6.3 and except in each case such termination shall not relieve (i) any party of any liability for any willful and material breach of this Agreement or (ii) Buyer of any liability for any breach of Section 5.4; and

(b) all filings, applications and other submissions made pursuant to this Agreement, to the extent practicable, shall be withdrawn from the agency or other person to which they were made.

## ARTICLE IX INDEMNIFICATION

## 9.1 <u>Survival</u>.

(a) The representations and warranties of the parties contained in Articles IV and V shall, subject to Section 9.1(c), survive the Closing until the date that is six (6) months after the Closing Date; <u>provided</u>, <u>however</u>, that any claim arising from a breach of Seller's representations or warranties set forth in Sections 4.2 (Authority), 4.4 (Capitalization) or 4.19 (Taxes) or Buyer's representations or warranties set forth in Section 5.2 (Authority) shall survive until the date that is ninety (90) days following the expiration of the applicable statute of limitations (including extensions thereof).

(b) All covenants and agreements contained herein that by their terms are to be performed in whole or in part, or which prohibit actions, subsequent to the Closing, shall survive the Closing in accordance with their terms. All other covenants and agreements contained herein shall not survive the Closing and shall thereupon terminate.

(c) The period of time a representation or warranty or covenant or agreement survives the Closing pursuant to this Section 9.1 shall be the "**Survival Period**" with respect to such representation or warranty or covenant or agreement. In the event notice of any claim for indemnification under this Article IX shall have been given within the applicable Survival Period and such claim has not been finally resolved by the

expiration of such Survival Period, the representations or warranties or covenants or agreements that are the subject of such claim shall survive until such claim is finally resolved.

# 9.2 <u>Indemnification Obligations of Seller</u>.

(a) From and after the Closing, subject to the terms of this Article IX, Seller shall indemnify and hold harmless Buyer and its Affiliates (other than the Partnership) and their respective directors, officers, employees, stockholders, partners, members, agents, attorneys, representatives, successors and assigns (collectively, the "**Buyer Indemnified Parties**"), from and against any losses, damages, liabilities, costs and expenses (including reasonable costs of investigation and reasonable attorneys' fees), interest, penalties, judgments, settlements, claims, obligations, deficiencies, demands, damages, fines, claims, suits, actions, causes of action, assessments, awards, whether or not involving a third party claim (collectively, "**Losses**") incurred by any Buyer Indemnified Party by reason of (i) any breach of any of the representations or warranties (in each case, when made) of Seller in Article IV and (ii) any breach of any of the covenants or agreements of Seller in this Agreement that by their terms are to be performed in whole or in part, or which prohibit actions, subsequent to the Closing Date.

(b) Except with respect to Losses incurred by any Buyer Indemnified Party as a result of a breach of Seller's representations or warranties set forth in Section 4.19 (Taxes), the obligation of Seller to indemnify any Buyer Indemnified Party for Losses is subject to the following limitations: (i) no Buyer Indemnified Party shall be entitled to make a claim against Seller for indemnification under Section 9.2(a)(i) ("**Buyer Claim**") unless and until the aggregate amount of Losses incurred by such Buyer Indemnified Party with respect to an event or occurrence and all other events or occurrences arising from the same circumstances exceeds \$75,000 (a "**Base Claim**"); (ii) Seller shall not be required to provide indemnified Parties in respect of Buyer Indemnified Party pursuant to Section 9.2(a)(i) unless the aggregate amount of Losses incurred by all the Buyer Indemnified Parties in respect of Buyer Claims constituting Base Claims exceeds 2.5% of the Closing Consideration (the "**Basket**"), and then the Buyer Indemnified Parties shall be entitled to indemnification for only the amount in excess of the Basket; and (iii) in no event shall the aggregate amount of Losses for which Seller is obligated to indemnify the Buyer Indemnified Parties pursuant to Section 9.2(a)(i) exceed 10% of the Closing Consideration (the "**Ceiling**").

# 9.3 <u>Indemnification Obligations of Buyer</u>.

(a) From and after the Closing, subject to the terms of this Article IX, Buyer shall indemnify and hold harmless Seller and its Affiliates (other than the Partnership) and their respective directors, officers, employees, stockholders, partners, members, agents, attorneys, representatives, successors and assigns (collectively, the "**Seller Indemnified Parties**") from and against Losses incurred by any Seller Indemnified Party by reason of (i) any breach of any of the representations or warranties (in each case, when made) of Buyer in Article V and (ii) any breach in any material respect of any of the covenants or agreements of Buyer in this Agreement that by their

34

terms are to be performed in whole or in part, or which prohibit actions, subsequent to the Closing Date.

(b) The obligation of Buyer to indemnify any Seller Indemnified Party for Losses is subject to the following limitations: (i) no Seller Indemnified Parties shall be entitled to make a claim against Buyer for indemnification under Section 9.3(a)(i) ("**Seller Claim**") unless and until the aggregate amount of Losses incurred by the Seller Indemnified Parties with respect to an event or occurrence and all other events or occurrences caused by the same circumstances constitutes a Base Claim; (ii) Buyer shall not be required to provide indemnification to any Seller Indemnified Party pursuant to Section 9.3(a) (i) ("**Seller Claims**") unless the aggregate amount of Losses incurred by all the Seller Indemnified Parties in respect of Seller Claims constituting Base Claims exceeds the Basket, and then the Seller Indemnified Parties shall be entitled to indemnification for only the amount in excess of the Basket; and (iii) in no event shall the aggregate amount of Losses for which Buyer is obligated to indemnify the Seller Indemnified Parties pursuant to Section 9.3(a)(i) of this Agreement exceed the Ceiling.

# 9.4 <u>Indemnification Procedures and Limitations</u>.

In the event that any action, suit, claim or proceeding is commenced by a third party involving a claim for which a party required (a) to provide indemnification hereunder (an "Indemnifying Party") may be liable to a party entitled to indemnification (an "Indemnified Party") hereunder (a "Third Party Claim"), the Indemnified Party shall promptly notify the Indemnifying Party in writing of such Third Party Claim indicating the nature of such claim and the basis therefore (the "Claim Notice"); provided, however, that no delay on the part of the Indemnified Party in giving any such Claim Notice shall relieve the Indemnifying Party of any indemnification obligation hereunder except to the extent that the Indemnifying Party is adversely prejudiced by such delay. Thereafter, the Indemnified Party shall deliver to the Indemnifying Party, promptly following the Indemnified Party's receipt thereof, copies of all notices and documents (including court papers) received by the Indemnified Party relating to the Third-Party Claim. Subject to the provisions related to the settlement of Third-Party Claims set forth in Section 9.4(b), the Indemnifying Party will have the right to defend against, negotiate, settle or otherwise deal with any Third-Party Claim and to select counsel of its choice. If the Indemnifying Party does not, within 30 days from its receipt of the Claim Notice (the "Notice Period"), elect to undertake to defend against, negotiate, settle or otherwise deal with any Third Party Claim, the Indemnified Party may, subject to the provisions related to the settlement of Third-Party Claims set forth in Section 9.4(b), defend against, negotiate, settle or otherwise deal with such Third Party Claim. If the Indemnifying Party undertakes to defend against such Third Party Claim, (i) the Indemnifying Party shall not be liable to the Indemnified Party for any legal fees or expenses incurred by the Indemnified Party in connection with such Third Party Claim and (ii) the Indemnified Party may participate and retain counsel, at its own cost and expense, in the defense of such Third Party Claim (it being understood and agreed that the Indemnifying Party shall control such defense); provided, however, the Indemnified Party will be entitled to participate in any such defense with separate counsel the reasonable fees and expenses of which the Indemnifying Party shall bear if, but only if,

(A) so requested by the Indemnifying Party to participate or (B) a conflict of interest exists between the Indemnified Party and the Indemnifying Party that would make it inappropriate in the reasonable judgment of such Indemnified Party (upon and in conformity with the advice of counsel) for the same counsel to represent both the Indemnified Party and the Indemnifying Party; <u>provided</u>, <u>further</u>, that the Indemnifying Party will not be required to pay the reasonable fees and expenses of more than one (1) such counsel for all Indemnified Parties in connection with any Third-Party Claim.

(b) If the Indemnifying Party undertakes to defend against such Third Party Claim, the other party shall (and shall cause the applicable Indemnified Parties to) (i) fully cooperate with the Indemnifying Party and its counsel in the investigation, defense and settlement of such Third Party Claim, including providing all information and documents available to the Indemnified Party which relate to such Third Party Claim and (ii) consent and agree to any settlement, compromise or discharge of a Third-Party Claim that the Indemnified Party in connection with such Third-Party Claim and (C) does not otherwise have a material and adverse effect on the Business as determined by the Indemnified Party in its reasonable discretion. If the Indemnifying Party elects not to undertake within the Notice Period to defend against a Third-Party Claim, then the Indemnified Party shall not admit any liability with respect to, or settle, compromise or discharge, such Third-Party Claim without the Indemnifying Party's prior written consent (which shall not be unreasonably withheld or delayed).

(c) In calculating amounts payable to an Indemnified Party, the amount of any indemnified Losses shall be determined without duplication of any other Loss for which an indemnification claim has been made or could be made under any other representation, warranty, covenant, or agreement and shall be computed net of (i) payments recovered by the Indemnified Party under any insurance policy with respect to such Losses (it being understood and agreed by the parties that Seller (in the case of Seller Indemnified Parties) and Buyer (in the case of Buyer Indemnified Parties) shall use their commercially reasonable efforts to effect any such recovery), (ii) any prior or subsequent recovery by the Indemnified Party from any Person with respect to such Losses and (iii) any Tax benefit receivable by the Indemnified Party with respect to such Losses.

(d) Notwithstanding any other provision of this Agreement, in no event shall any Indemnified Party be entitled to indemnification pursuant to this Article IX (i) to the extent such Indemnified Party has otherwise been compensated for such Losses by reason of any adjustment contemplated by Section 2.3 or (ii) to the extent any Losses were attributable to such Indemnified Party's own gross negligence or willful misconduct.

(e) To the extent that Seller makes any payment pursuant to this Article IX in respect of Losses for which Buyer or any of its Affiliates have a right to recover against a third party (including an insurance company), Seller shall be subrogated to the right of Buyer or any of its Affiliates to seek and obtain recovery from such third

36

party; <u>provided</u>, <u>however</u>, that if Seller shall be prohibited from such subrogation, Buyer shall use its commercially reasonable efforts to seek recovery from such third party on Seller's behalf and shall pay any such recovery to Seller.

(f) Notwithstanding any other provision of this Agreement, in no event shall Seller or Buyer be liable for (i) punitive damages or any special, incidental, indirect or consequential damages of any kind or nature, regardless of the form of action through which such damages are sought or (ii) lost profits or diminution in value resulting from a breach or an alleged breach of any representation, warranty, covenant or other agreement set forth in this Agreement or otherwise in connection with the transactions contemplated hereby, even if under applicable Law, such lost profits or diminution in value would not be considered consequential or special damages.

(g) Except as otherwise provided by applicable law, Buyer and Seller agree to treat any indemnity payment made pursuant to this Article IX as an adjustment to the Purchase Price paid to Seller for all income Tax purposes.

(h) The remedies provided in this Article IX shall be the sole and exclusive remedies of the parties, from and after the Closing Date, with respect to this Agreement and the transactions contemplated hereby.

## ARTICLE X

## MISCELLANEOUS PROVISIONS

10.1 <u>Amendment and Modification</u>. Subject to applicable law, this Agreement may be amended, modified or supplemented only by written agreement of Seller and Buyer.

10.2 <u>Waiver of Compliance; Consents</u>. Except as otherwise provided in this Agreement, any failure of any of the parties to comply with any obligation, covenant, agreement or condition herein may be waived by the party entitled to the benefits thereof only by a written instrument signed by the party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

10.3 <u>Notices</u>. All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or by facsimile transmission, or mailed by a nationally recognized overnight courier or registered or certified mail (return receipt requested), postage prepaid, to the parties at the following addresses (or at such other address for a party as shall be specified by like notice, provided that notices of a change of address shall be effective only upon receipt thereof):

(a) If to Seller, to:

35

Tuscarora Gas Pipeline Co. 6100 Neil Road Reno, Nevada 89511 Attention: Michael Yackira Telecopy: (702) 367-5300

with copies (which shall not constitute notice) to:

Skadden, Arps, Slate, Meagher & Flom LLP 4 Times Square New York, New York 10036-6522 Attention: Sheldon S. Adler, Esq. Telecopy: (212) 735-2000

(b) if to Buyer, to:

TC PipeLines GP, Inc. 450 1<sup>st</sup> Street S.W. Calgary, Alberta, Canada T2P 5H1 Attention: Secretary Telecopy: (403) 920-2460

with copies (which shall not constitute notice) to:

TransCanada PipeLines Limited 450 1<sup>st</sup> Street S.W. Calgary, Alberta, Canada T2P 5H1 Attention: General Counsel Telecopy: (403) 920-2410

10.4 <u>Assignment</u>. This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto, including by operation of law, without the prior written consent of the other party (which may be withheld in its sole discretion); <u>provided</u>, that, without the consent of Seller, but with prior notice to Seller, Buyer may assign all or any portion of its rights and obligations under this Agreement to one or more Affiliates of Buyer (provided, however, that (a) each such assignee shall execute a supplementary agreement pursuant to which it agrees to be bound by the terms and conditions of this Agreement and (b) no assignment of rights or obligations pursuant to this Section 10.4 shall relieve the assigning party of its obligations hereunder).

10.5 Arbitration.

(a) Any dispute, controversy or claim arising out of or relating to this Agreement or the Interest Holder Agreement, or the breach, termination or validity hereof or the transactions contemplated hereby or thereby ("**Dispute**"), shall be finally

38

settled by arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association (the "AAA") then in effect (the "Rules").

(b) The place of arbitration shall be New York, NY. If the amount in controversy is one million dollars (\$1,000,000) or less (including all claims and counterclaims) there shall be one arbitrator who shall be agreed upon by the parties within 20 days of receipt by respondent of a copy of the demand for arbitration. If the amount in controversy is more than one million dollars (\$1,000,000) (including all claims and counterclaims), there shall be three (3) neutral and impartial arbitrators, of whom one (1) shall be appointed by Buyer and one (1) shall be appointed by Seller within 30 days of receipt by respondent(s) of the demand for arbitration. The two (2) arbitrators so appointed shall appoint a third arbitrator within 20 days of the appointment of the second arbitrator, who shall serve as the chair of the arbitral tribunal. If any arbitrator is not appointed within the time limits provided herein, such arbitrator shall be appointed by the AAA in accordance with the listing, striking and ranking procedure in the Rules. Any arbitrator appointed by the AAA shall be a retired judge or a practicing attorney with no less than 15 years of experience with large commercial cases.

(c) Any arbitration conducted pursuant hereto shall be confidential. No party or any of its agents shall disclose or permit the disclosure of any information about the evidence adduced or the documents produced by the other in the arbitration proceedings or about the existence, contents or results of the proceedings except as may be required by a Governmental Authority or as required in an action in aid of arbitration or for enforcement of an arbitral award. Before making any disclosure permitted by the preceding sentence, the party intending to make such disclosure shall give the other party reasonable written notice of the intended disclosure and afford the other party a reasonable opportunity to protect their interests.

(d) In rendering an award, the arbitral tribunal shall be required to follow the laws of the State of New York. The arbitral tribunal is not empowered to award damages in excess of compensatory damages, and each party hereby irrevocably waives any right to recover punitive, exemplary or similar damages with respect to any Dispute. Any arbitration proceedings, decision or award rendered hereunder and the validity, effect and interpretation of this arbitration agreement shall be governed by the Federal Arbitration Act, 9 U.S.C. §1 et seq. The award shall be in writing and shall briefly state the findings of fact and conclusions of applicable laws on which it is based. The award shall be final and binding upon the parties and shall be the sole and exclusive remedy between the parties regarding any claims, counterclaims, issues or accounting presented to the arbitrator(s). Judgment upon the award may be entered in any court having jurisdiction. Any costs or fees (including attorneys' fees and expenses) incident to enforcing the award shall be charged against the party resisting such enforcement.

(e) By agreeing to arbitration, the parties do not intend to deprive any court of its jurisdiction to issue a pre-arbitral injunction, prearbitral attachment, or other order in aid of arbitration proceedings and the enforcement of any award. Without prejudice to such provisional remedies as may be available under the jurisdiction of a court, the arbitral tribunal shall have full authority to grant provisional remedies and to

39

direct the parties to request that any court modify or vacate any temporary or preliminary relief issued by such court, and to award damages for the failure of any party to respect the arbitral tribunal's orders to that effect. The parties hereby submit to the exclusive jurisdiction of the federal and state courts located in New York, New York (the "**New York Courts**") for the purpose of an order to compel arbitration, for preliminary relief in aid of arbitration or for a preliminary injunction to maintain the status quo or prevent irreparable harm prior to the appointment of the arbitrators, and to the non-exclusive jurisdiction of the New York Courts for the enforcement of any award issued hereunder. Each party unconditionally and irrevocably waives any objections which they may have now or in the future to such jurisdiction including objections by reason of lack of personal jurisdiction, improper venue, or inconvenient forum.

(f) The parties are committed to the prompt and efficient resolution of disputes. Accordingly, if one or more disputes arise under this Agreement, such disputes may be brought in a single arbitration. Upon the request of any party, any dispute or arbitration proceeding constituted under this Agreement shall be consolidated with any other dispute or arbitration proceeding constituted under this Agreement. The new dispute or arbitration shall be so consolidated, provided that the arbitral tribunal for the prior (or first filed) arbitration determines that: (i) the new dispute or arbitration presents significant issues of applicable law or fact common with those in the pending arbitration; (ii) no party would be unduly prejudiced by such consolidation; and (iii) consolidation under such circumstances would not result in undue delay for the prior arbitration. Any order of consolidation issued by the arbitral tribunal shall be final and binding upon the parties. The arbitral tribunal for the arbitration constituted first in time shall serve as the arbitrat tribunal for the consolidation under the parties waive any right they have to appeal or to seek interpretation, revision or annulment of such order of consolidation under the Rules or in any court. The parties agree that upon such an order of consolidation, they will promptly dismiss any arbitration brought under this Agreement, the subject of which has been consolidated into another arbitral proceeding.

10.6 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of New York (regardless of the laws that might otherwise govern under applicable New York principles of conflicts of law) as to all matters, including matters of validity, construction, effect, performance and remedies.

10.7 <u>Facsimiles; Counterparts</u>. This Agreement may be executed by facsimile signatures by any party and such signature shall be deemed binding for all purposes hereof, without delivery of an original signature being thereafter required. This Agreement may be executed in one or more counterparts, each of which, when executed, shall be deemed to be an original and all of which together shall constitute one and the same document.

10.8 <u>Interpretation</u>. The article and section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties and shall not in any way affect the meaning or interpretation of this Agreement. When a reference is made in this Agreement to an Article, Section, Exhibit or Schedule,

40

such reference shall be to an Article, Section, Exhibit or Schedule of or to this Agreement unless otherwise indicated. References in this Agreement to "dollars" or "\$" are to U.S. dollars. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." When a reference in this Agreement is made to a "party" or "parties," such reference shall be to a party or parties to this Agreement unless otherwise indicated.

10.9 <u>Entire Agreement</u>. This Agreement constitutes the entire agreement and understanding of the parties with respect to the subject matter hereof and supersedes all prior agreements and understandings between the parties with respect to such subject matter; <u>provided</u> that this Agreement shall not supersede the terms and provisions of the Confidentiality Agreement, which shall survive and remain in effect until the expiration or termination thereof in accordance with its terms.

10.10 <u>Severability</u>. Whenever possible, each provision or portion of any provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law but if any provision or portion of any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability will not affect any other provision or portion of any provision in such jurisdiction, and this Agreement will be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision or portion of any provision had never been contained herein.

10.11 <u>Third Party Beneficiaries</u>. This Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their respective successors and assigns. None of the provisions of this Agreement shall be for the benefit of or enforceable by any third party, including any creditor of any party or any of their Affiliates. No such third party shall obtain any right under any provision of this Agreement or shall by reasons of any such provision make any claim in respect of any liability (or otherwise) against either party hereto.

IN WITNESS WHEREOF, Seller and Buyer have caused this Agreement to be signed by their respective duly authorized officers as of the date first above written.

By:	/s/ Michael W. Yackira
Name:	Michael W. Yackira
Title:	Treasurer
TC TUSCARORA INTERMEDIATE LIMITED PARTNERSHIP	
By: TC P general pa	ipeLines GP, Inc., its artner
By:	/s/ Mark Zimmerman
Name:	Mark Zimmerman
Title:	VP – Business Development
By: <u>/s/</u>	Ron Cook
Name:	Ron Cook
Title:	Vice President - Taxation

# TABLE OF CONTENTS

		Page
ARTICLE I DEF	INITIONS	1
1.1	Definitions	1
		10
	RCHASE AND SALE OF GENERAL PARTNERSHIP INTEREST	10
2.1	Purchase and Sale	10
2.2	Purchase Price	10
2.3	Purchase Price Adjustment	11
ARTICLE III TH	E CLOSING	14
3.1	Time and Place of Closing	14
3.2	Deliveries by Seller	14
3.3	Deliveries by Buyer	15
ARTICLE IV RE	PRESENTATIONS AND WARRANTIES OF SELLER	15
4.1	Organization; Qualification	15
4.2	Authority Relative to this Agreement and Related Agreements	16
4.3	Consents and Approvals; No Violation	16
4.4	Capitalization	17
4.5	Financial Statements	17
4.6	Undisclosed Liabilities	17
4.7	Absence of Certain Changes or Events	17
4.8	Compliance with Law	17
4.9	Title to Real Property	18
4.10	Intellectual Property	18
4.11	Material Leases	18
4.12	Insurance	18
4.13	Environmental Matters	19
4.14	Employee Matters	19
4.15	Material Contracts	19
4.16	Legal Proceedings, etc	21
4.17	Permits	21
4.18	Taxes	22
4.19	Brokers' Fee	22
ARTICLE V REI	PRESENTATIONS AND WARRANTIES OF BUYER	23
5.1	Organization	23
5.2	Authority Relative to this Agreement	23
5.3	Consents and Approvals; No Violation	23
5.4	Availability of Funds	23
5.5	Acquisition for Investment; Due Diligence	24
5.6	Brokers' Fee	24
0.0		

6.1	Conduct of Business of the Partnership	24
6.2	Access to Information	26
6.3	Expenses	27
6.4	Further Assurances	27
6.5	Public Statements	28
6.6	Consents and Approvals	28
6.7	Tax Matters	29
6.8	Supplements to Schedules	29
6.9	No Implied Representations	29
6.10	Financial Statements	30
ARTICLE VII C	LOSING CONDITIONS	30
7.1	Conditions to Each Party's Obligations to Effect the Transactions Contemplated Hereby	30
7.2	Conditions to Obligations of Buyer	31
7.3	Conditions to Obligations of Seller	31
ARTICLE VIII	TERMINATION	32
8.1	Termination	32
8.2	Procedure and Effect of Termination	33
ARTICLE IX IN	DEMNIFICATION	33
9.1	Survival	33
9.2	Indemnification Obligations of Seller	34
9.3	Indemnification Obligations of Buyer	34
9.4	Indemnification Procedures and Limitations	35
ARTICLE X MI	SCELLANEOUS PROVISIONS	37
10.1	Amendment and Modification	37
10.2	Waiver of Compliance; Consents	37
10.3	Notices	37
10.4	Assignment	38
10.5	Arbitration	38
10.6	Governing Law	40
10.7	Facsimiles; Counterparts	40
10.8	Interpretation	40
10.9	Entire Agreement	41
10.10	Severability	41
10.11	Third Party Beneficiaries	41

#### GENERAL PARTNER INTEREST HOLDER AGREEMENT

#### BY AND BETWEEN

#### TUSCARORA GAS PIPELINE CO.

#### AND

#### TC TUSCARORA INTERMEDIATE LIMITED PARTNERSHIP

November 1, 2006

#### **TABLE OF CONTENTS** Page ARTICLE I DEFINITIONS Section 1.1 Definitions 3 ARTICLE II COVENANTS Section 2.1 Certain Restrictions 3 Section 2.2 Option to Purchase TGPC Interest Δ Section 2.3 Voting Agreement 5 ARTICLE III TERM Section 3.1 Term 5 ARTICLE IV MISCELLANEOUS Section 4.1 Dispute Resolution 6 Section 4.2 Entire Agreement 6 Section 4.3 Expenses 6 Section 4.4 Amendment 6 Section 4.5 Notices 6 6 Section 4.6 Severability 7 Section 4.7 Waiver 7 Section 4.8 Interpretation 7 Section 4.9 Third Party Beneficiaries 7 Section 4.10 Assignment 8 Section 4.11 Governing Law Section 4.12 Facsimiles; Counterparts 8

## GENERAL PARTNER INTEREST HOLDER AGREEMENT

This GENERAL PARTNER INTEREST HOLDER AGREEMENT (this "**Agreement**"), dated as of November 1, 2006, is made by and between Tuscarora Gas Pipeline Co., a Nevada corporation ("**TGPC**"), and TC Tuscarora Intermediate Limited Partnership, a Delaware limited partnership ("**Buyer**").

### WITNESSETH:

WHEREAS, simultaneously with the execution of this Agreement, TGPC and Buyer, are entering into that certain General Partnership Interest Purchase Agreement, dated as of the date hereof (the "**Purchase Agreement**"), pursuant to which Buyer will acquire a 49% general partnership interest in Tuscarora Gas Transmission Company, a Nevada general partnership (the "**Partnership**");

WHEREAS, TGPC will retain a 1% general partnership interest in the Partnership (the "TGPC Interest"); and

WHEREAS, the parties desire to set forth their agreement as to certain rights and obligations relating to the TGPC Interest upon the terms and subject to the conditions hereinafter set forth;

NOW, THEREFORE, in consideration of the foregoing and the mutual premises, representations, warranties, covenants and agreements contained herein and in the Purchase Agreement, the parties hereto, intending to be legally bound hereby, agree as follows:

#### ARTICLE I

## DEFINITIONS

Section 1.1 <u>Definitions</u> For purposes of this Agreement, capitalized terms used, but not defined herein, shall have the meaning set forth in the Purchase Agreement.

#### ARTICLE II

#### **COVENANTS**

Section 2.1 Certain Restrictions.

(a) TGPC and Buyer agree that, without the prior written consent of Buyer, TGPC will not, directly or indirectly:

(i) sell, exchange, lend, transfer, assign, offer, pledge, or otherwise dispose of all or any portion of the TGPC

Interest;

(ii) sell or grant any option, right, warrant or contract to purchase or sell all or any portion of the TGPC Interest;

(iii) enter into any swap or other agreement that transfers, in whole or in part, the economic consequence of ownership of the TGPC Interest; and

(iv) make any proposal or publicly disclose any intention, plan or arrangement inconsistent with the foregoing.

(b) Neither Buyer nor TGPC shall disclose the terms, or existence, of this Agreement to any Person except as may be permitted by the Purchase Agreement.

Section 2.2 Option to Purchase TGPC Interest.

(a) Upon the Closing, Buyer shall pay \$100,000 to TGPC.

(b) Buyer shall have the right, but not the obligation, to purchase the TGPC Interest at any time during the period beginning from and after the closing of the transactions contemplated by the Purchase Agreement and ending on the date that is 30 days after the twelve-month anniversary of the closing of the transactions contemplated by the Purchase Agreement (the "**Option Expiration Date**") at a price in cash equal to the sum of (i) (A) the Purchase Price, <u>divided by</u> (B) 0.98, <u>multiplied by</u> (C) 0.02, <u>plus</u> (ii) the aggregate amount of any capital contributions in respect of the TGPC Interest paid by TGPC from and after the Closing and prior to the Option Closing (as defined below), <u>minus</u> (iii) the aggregate amount of any distributions paid by the Partnership to TGPC in respect of the TGPC Interest from and after the Closing and prior to the Option Closing, <u>minus</u> (iv) \$100,000.

(c) The closing of the transaction contemplated by this Section 2.2 (the "**Option Closing**") shall take place at the offices of Skadden, Arps, Slate, Meagher & Flom LLP, 4 Times Square, New York, New York, at 10:00 a.m., local time, no later than the third Business Day following the date on which written notice of Buyer's exercise of the option contemplated by this Section 2.2 is delivered to TGPC, or at such other place or time as the parties may agree.

(d) The respective obligations of each party to effect the transactions contemplated hereby to occur at the Option Closing shall be subject to the expiration or termination of the waiting period under the HSR Act applicable to the consummation of the transactions contemplated hereby. In addition, it shall be a condition to Buyer's exercise of its right to purchase the TGPC Interest hereunder (and TGPC's obligation to sell the TGPC Interest hereunder) that the Partnership, Buyer and TCPL consent to such disposition and waive any claims involving a breach of the Partnership Agreement or a termination of the Partnership (whether under Section 708 of the Code or otherwise) in connection therewith (pursuant to documentation reasonably acceptable to TGPC).

(e) TGPC shall not be required to make any representations and warranties or provide any indemnification or otherwise incur any obligations in connection with the transactions contemplated hereby, except:

(i) (A) the TGPC Interest has been duly authorized and validly issued, and is fully paid and nonassessable, and is held of record and beneficially owned by TGPC; and (B) at the Option Closing (assuming the payment of the purchase price contemplated by Section 2.2(b)), TGPC will transfer the TGPC Interest to Buyer, free and clear of Encumbrances (other than Encumbrances created by Buyer or arising under applicable securities laws), which representations and warranties shall be true and correct (A) as of the date hereof, (B) as of the Closing and (C) as of the Option Closing; and

(ii) the indemnification obligations set forth in Article IX of the Purchase Agreement for breaches of the representations and warranties in Section 2.2(e)(i) shall apply to the TGPC Interest from and after the Closing (subject to the limitations and other provisions set forth in Article IX of the Purchase Agreement).

## Section 2.3 Voting Agreement.

TGPC agrees to vote, or cause to be voted, the TGPC Interest, from time to time and at all times any vote of the Partners (as defined in the Partnership Agreement) is required under the Partnership Agreement in such manner as directed by Buyer until this Agreement is terminated pursuant to Section 3.1.

#### TERM

#### Section 3.1 Term.

This Agreement shall terminate automatically (a) upon consummation of the sale of the TGPC Interest by TGPC to Buyer pursuant to Section 2.2, (b) upon the expiration without exercise by Buyer of its right to purchase the TGPC Interest described in Section 2.2 or (c) upon the termination of the Purchase Agreement in accordance with its terms. If this Agreement is terminated as provided herein, such termination shall be the sole remedy of the parties hereto with respect to breaches of this Agreement and none of the parties hereto nor any of their respective trustees, directors, officers, members, general partners, or Affiliates, as the case may be, shall have any liability or further obligation to the other party or any of their respective trustees, directors, officers, members, general partners, or Affiliates, as the case may be, pursuant to this Agreement, except in each case as provided in Section 4.3.

#### ARTICLE IV

#### MISCELLANEOUS

#### Section 4.1 Dispute Resolution.

Any disputes arising out of or relating to this Agreement or the breach, termination or validity thereof or the parties' performance hereunder shall be resolved as provided by Section 10.5 of the Purchase Agreement.

#### Section 4.2 Entire Agreement.

This Agreement constitutes the entire agreement and understanding of the parties with respect to the subject matter hereof and supersedes all prior agreements and understandings between the parties with respect to such subject matter; <u>provided</u> that this Agreement shall not supersede the terms and provisions of the Confidentiality Agreement, which shall survive and remain in effect until expiration or termination thereof in accordance with its terms and this Agreement.

#### Section 4.3 Expenses.

Except to the extent otherwise specifically provided herein, whether or not the transactions contemplated hereby are consummated, all costs and expenses incurred in connection with this Agreement and the transactions contemplated hereby shall be borne by the party incurring such costs and expenses.

#### Section 4.4 Amendment.

Subject to applicable law, this Agreement may be amended, modified or supplemented only by written agreement of TGPC and Buyer.

#### Section 4.5 Notices.

All notices and other communications hereunder shall be in writing and shall be deemed given if delivered personally or by facsimile transmission, or mailed by a nationally recognized overnight courier or registered or certified mail (return receipt requested), postage prepaid, to the parties at the addresses set forth in Section 10.3 of the Purchase Agreement (or at such other address for a party as shall be specified by like notice, provided that notices of a change of address shall be effective only upon receipt thereof).

#### Section 4.6 Severability.

Whenever possible, each provision or portion of any provision of this Agreement will be interpreted in such manner as to be effective and valid under applicable law but if any provision or portion of any provision of this Agreement is held to be invalid, illegal or unenforceable in any respect under any applicable law or rule in

any jurisdiction, such invalidity, illegality or unenforceability will not affect any other provision or portion of any provision in such jurisdiction, and this Agreement will be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision or portion of any provision had never been contained herein.

#### Section 4.7 Waiver.

Except as otherwise provided in this Agreement, any failure of any of the parties to comply with any obligation, covenant, agreement or condition herein may be waived by the party entitled to the benefits thereof only by a written instrument signed by the party granting such waiver, but such waiver or failure to insist upon strict compliance with such obligation, covenant, agreement or condition shall not operate as a waiver of, or estoppel with respect to, any subsequent or other failure.

#### Section 4.8 Interpretation.

The article and section headings contained in this Agreement are solely for the purpose of reference, are not part of the agreement of the parties and shall not in any way affect the meaning or interpretation of this Agreement. When a reference is made in this Agreement to a Section, such reference shall be to a Section of this Agreement unless otherwise indicated. References in this Agreement to "dollars" or "\$" are to U.S. dollars. Whenever the words "include," "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation." When a reference in this Agreement is made to a "party" or "parties," such reference shall be to a party or parties to this Agreement unless otherwise indicated.

#### Section 4.9 Third Party Beneficiaries.

This Agreement shall be binding upon and inure solely to the benefit of the parties hereto and their respective successors and assigns. None of the provisions of this Agreement shall be for the benefit of or enforceable by any third party, including any creditor of any party or any of their affiliates. No such third party shall obtain any right under any provision of this Agreement or shall by reasons of any such provision make any claim in respect of any liability (or otherwise) against either party hereto.

## Section 4.10 Assignment.

This Agreement and all of the provisions hereof shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns, but neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned by any party hereto, including by operation of law without the prior written consent of the other party (which may be withheld in its sole discretion); provided, that, without the consent of TGPC, but with prior notice to TGPC, Buyer may assign all or any portion of its rights and obligations under this Agreement to one or more Affiliates of Buyer (provided, however, that (a) each such assignee shall execute a supplementary agreement pursuant to which it agrees to be bound by the terms and

conditions of this Agreement and (b) no assignment of rights or obligations pursuant to this Section 4.10 shall relieve the assigning party of its obligations hereunder).

#### Section 4.11 Governing Law.

This Agreement shall be governed by and construed in accordance with the laws of the State of New York (regardless of the laws that might otherwise govern under applicable New York principles of conflicts of law) as to all matters, including matters of validity, construction, effect, performance and remedies.

#### Section 4.12 Facsimiles; Counterparts.

This Agreement may be executed by facsimile signatures by any party and such signature shall be deemed binding for all purposes hereof, without delivery of an original signature being thereafter required. This Agreement may be executed in one or more counterparts, each of which, when executed, shall be deemed to be an original and all of which together shall constitute one and the same document.

IN WITNESS WHEREOF, TGPC and Buyer have caused this Agreement to be signed by their respective duly authorized officers as of the date first above written.

#### TUSCARORA GAS PIPELINE CO.

By: /s/ Michael W. Yackira

Name:	Michael W. Yakira
Title:	Treasurer

#### TC TUSCARORA INTERMEDIATE LIMITED PARTNERSHIP

By: TC PipeLines GP, Inc., its general partner

By: /s/ Mark Zimmerman

Name:	Mark Zimmerman
Title:	VP – Business Development

By: /s/ Ron Cook

Name: Ron Cook Title: Vice President - Taxation

Exhibit 99.1

Press Release: TC PipeLines, LP to acquire controlling interest in Interstate Pipeline System

November 2, 2006

Calgary, Alberta – November 2, 2006 – (Nasdaq: TCLP) – TC PipeLines, LP (the Partnership) today announced it has signed agreements to purchase Sierra Pacific Resources' 50 per cent interest in the Tuscarora Gas Transmission Company (Tuscarora) for \$100 million. The Partnership will also indirectly assume \$37 million of Tuscarora debt. Upon completion of the transaction, TC PipeLines, LP will own or control 99 per cent of Tuscarora. TransCanada Corporation (TransCanada), the parent company of TC PipeLines GP, Inc. and the sole general partner of TC PipeLines, LP, indirectly holds the remaining 1 per cent ownership interest. The transaction is expected to close by year end, subject to regulatory approvals.

TransCanada will become the operator of Tuscarora, which is currently operated by an affiliate of Sierra Pacific Resources. TransCanada has provided gas control services for the Tuscarora pipeline system since late 2002.

Tuscarora has firm transportation contracts for over 95 per cent of its available contracted capacity. A subsidiary of Sierra Pacific Resources holds contracts representing 69 per cent of the total available capacity, the majority of which expire on October 31, 2017.

"TC PipeLines, LP is focused on delivering stable, sustainable cash flows and finding growth opportunities and we're pleased to continue to deliver on that goal," said Russ Girling, president and chief executive officer of TC PipeLines GP, Inc., general partner of TC PipeLines, LP. "The Tuscarora pipeline system is a high quality asset supported by long-term contracts."

This acquisition is expected to be accretive to the Partnership's cash flow based on anticipated financing with a combination of bank debt and equity.

Tuscarora owns a 240-mile, 20-inch diameter, United States interstate pipeline system with a subscribed capacity of approximately 180 MMcf/d. The Tuscarora pipeline system originates at an interconnection point with existing facilities of TransCanada's Gas Transmission Northwest System near Malin, Oregon and runs southeast through northeastern California and northwestern Nevada, terminating near Wadsworth, Nevada.

The Tuscarora acquisition follows on the Partnership's acquisition in April 2006 of an additional 20 per cent of Northern Border Pipeline Company, an interstate pipeline that transports natural gas from the Montana-Saskatchewan border to markets in the U.S. midwest. Northern Border Pipeline is a leading transporter of natural gas imported from Canada into the U.S.

## Analyst/Media Teleconference Dial-in and Webcast Information

The Partnership will hold a conference call Thursday, November 2, 2006 at 1 p.m. (Eastern). Russ Girling, president and chief executive officer of the general partner, will discuss the third quarter 2006 financial results and general developments and issues concerning the Partnership. Those interested in listening to the call may dial (866) 898-9626. A replay of the conference call will also be available two hours after the call and until midnight (Eastern), Thursday, November 9, 2006, by dialing (800) 408-3053, then entering passcode 3201008.

A live webcast of the conference call will also be available through the Partnership's website at www.tcpipelineslp.com. An audio replay of the call will be available on the website.

TC PipeLines, LP is a publicly traded limited partnership. It owns a 50 per cent interest in Northern Border Pipeline Company, a Texas general partnership, and a 49 per cent interest in Tuscarora Gas Transmission Company, a Nevada general partnership. Northern Border Pipeline owns a 1,249-mile United States interstate pipeline system that transports natural gas from the Montana-Saskatchewan border to markets in the midwestern United States. Tuscarora owns a 240-mile United States interstate pipeline system that transports natural gas from the Montana-Saskatchewan border to markets in the midwestern United States. Tuscarora owns a 240-mile United States interstate pipeline system that transports natural gas from Oregon where it interconnects to TransCanada's Gas Transmission Northwest System. TC PipeLines, LP is managed by its general partner, TC PipeLines GP, Inc., an indirect wholly owned subsidiary of TransCanada Corporation. TC PipeLines GP, Inc., also holds common units of the Partnership. Common units of TC PipeLines, LP are quoted on the NASDAQ Stock Market and trade under the symbol "TCLP."

## Cautionary Statement Regarding Forward-Looking Information

This news release may include forward-looking statements regarding future events and the future financial performance of TC PipeLines, LP. Words such as "believes," "expects," "intends," "forecasts," "projects," "goals," and similar expressions identify forward-looking statements. All forward-looking statements are based on the Partnership's current beliefs as well as assumptions made by and information currently available to the Partnership. These statements reflect the Partnership's current views with respect to future events. The Partnership assumes no obligation to update any such forward-looking statement to reflect events or circumstances occurring after the date hereof. Important factors that could cause actual results to materially differ from the Partnership's current expectations include completion of the acquisition of an additional 50 per cent interest in Tuscarora and related transactions, regulatory decisions, particularly those of the Federal Energy Regulatory Commission and the Securities and Exchange Commission, the ability of Northern Border Pipeline to recontract its available capacity at maximum rates, operational decisions of Northern Border Pipeline's operator, the failure of a shipper on either one of the Partnership's pipelines to perform its contractual obligations, cost of acquisitions, future demand for natural gas, overcapacity in the industry, and other risks inherent in the transportation of natural gas as discussed in the Partnership's filings with the Securities and Exchange Commission, including the Partnership's Annual Report on Form 10-K for the year ended December 31, 2005.

## **Media Inquiries:**

Jennifer Varey (403) 920-7859 or (800) 608-7859

## **Unitholder and Analyst Inquiries:**

Myles Dougan Toll-free (877) 290-2772 investor\_relations@tcpipelineslp.com